<SUBMISSION>
<TYPE>
<DOCUMENT-COUNT>
<LIVE>
<FILER-CIK>
<FILER-CCC>
<CONTACT-NAME>
<CONTACT-PHONE-NUMBER>
<SROS> 8-K 10 0000915781 ####### EDGAR FILING GROUP 312-707-9790 <CONTACT-PHONE-NOT
<SROS>
<PERIOD>
<NOTIFY-INTERNET>
<NOTIFY-INTERNET>
<ITEMS>
<ITEMS>
<ITEMS> NASD 09-20-2007 bowne.edgar@bowne.com csd.minneapolis@bowne.com 1.01 5.02 8.01 <ITEMS>
<ITEMS>

<TITEMS> 9.01 <ADDITIONAL-SUBMISSION-TYPE> DEFA14A

<DOCUMENT>
<TYPE> 8-K
<FILENAME> c18876e8vk.htm
<DESCRIPTION> Current Report
<TEXT>

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (date of earliest event reported): September 20, 2007

$\underbrace{RTW, Inc.}_{\text{(Exact name of Registrant as Specified in its Charter)}}$

Minnesota (State Or Other Jurisdiction Of Incorporation)

0-25508	41-1440970		
(Commission File Number) (I.R.S. Employer Identification I			
8500 Normandale Lake Blvd., Suite 1400			
Bloomington, MN	55437		
(Address Of Principal Executive Offices)	(Zip Code)		
Registrant's Telephone Nur Check the appropriate box below if the Form 8-K filing is intended to sit the following provisions (<i>see</i> General Instruction A.2. below):	mber, Including Area Code		
☐ Written communications pursuant to Rule 425 under the Securities A	ct (17 CFR 230.425)		
$\ensuremath{\square}$ Soliciting material pursuant to Rule 14a-12 under the Exchange Act ((17 CFR 240.14a-12)		
☐ Pre-commencement communications pursuant to Rule 14d-2(b) under	r the Exchange Act (17 CFR 240.14d-2(b))		
☐ Pre-commencement communications pursuant to Rule 13e-4(c) under	r the Exchange Act (17 CFR 240.13e-4(c))		

TABLE OF CONTENTS

Item 1.01 Entry Into a Material Definitive Agreement

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Officers;

Compensatory Arrangements of Certain Officers

Item 8.01 Other Events

Item 9.01 Financial Statements And Exhibits

SIGNATURE

Agreement and Plan of Merger

Amendment to and Termination of 1995 Employee Stock Purchase Plan

Amendment to and Termination of 1994 Stock Plan

Amendment to and Termination of 2005 Stock Plan

Form of Employment Agreement

Press Release

Quastion and Answer Fact Sheet

Items under Sections 2, 3, 4, 6 and 7 are not applicable and therefore omitted.

Item 1.01 Entry Into a Material Definitive Agreement.

On September 20, 2007, RTW, Inc., a Minnesota corporation (the "Company" or "RTW"), Rockhill Holding Company ("Rockhill") and Rockhill Acquisition Corporation, a Minnesota corporation and wholly owned subsidiary of Rockhill ("Merger Sub"), entered into an Agreement and Plan of Merger (the "Merger Agreement"). The Merger Agreement provides that, upon the terms and subject to the conditions set forth in the Merger Agreement, Merger Sub will merge with and into the Company (the "Merger"), with the Company continuing as the surviving corporation. Upon consummation of the Merger, the Company will become a wholly-owned subsidiary of Rockhill.

The Merger Agreement

On the terms and subject to the conditions of the Merger Agreement, at the effective time of the Merger, each share of Company common stock that is then outstanding (other than shares as to which dissenters' rights have been properly exercised) will be converted into the right to receive \$12.45 in cash. The Merger Agreement provides that all options to purchase Company common stock will automatically be accelerated and will become fully vested and exercisable immediately prior to the effective time of the Merger. To the extent not exercised immediately prior to the effective time, each outstanding option to purchase Company common stock will be cancelled and converted into the right to receive a cash amount equal to the excess, if any, of \$12.45 over the exercise price for such option.

The Company's board of directors has unanimously approved the Merger Agreement and has unanimously recommended that the Company's shareholders approve and adopt the Merger Agreement.

The Company and Rockhill have made customary representations, warranties and covenants in the Merger Agreement, including, among others: (i) covenants generally requiring the Company to conduct its business prior to the closing of the Merger in the ordinary course consistent with past practice; (ii) covenants restricting the solicitation of competing acquisition proposals; and (iii) covenants relating to obtaining the required approval of the Company's shareholders.

Consummation of the Merger is subject to customary conditions including, among other things: (i) approval by the Company's shareholders; (ii) expiration or early termination of the applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 and (iii) approval of appropriate state insurance regulatory authorities. In addition to customary conditions, the following additional conditions must be satisfied before Rockhill is obligated to consummate the Merger: (i) certain RTW executives must enter into employment agreements with Rockhill (the terms of these employment arrangements are summarized under Item 5.02 below); (ii) the Company's net ultimate liability for incurred loss and allocated loss adjustment expense (excluding inter-company loss adjusting expense) must not have materially adversely changed; (iii) the Company's contract with Minnesota Workers' Assigned Risk Plan must be in full force; (iv) the Company may not incur more than \$2 million in transaction-related expenses in the period between signing the Merger Agreement and the closing of the Merger; and (v) the states of Minnesota, Colorado and Michigan cannot in connection with approving the Merger require any material changes in the manner in which the Company currently conducts its business.

Under the Merger Agreement, each of the Company and Rockhill has certain rights to terminate the Merger Agreement and the Merger. Upon the termination of the Merger Agreement under certain specified circumstances, the Company must pay Rockhill a termination fee of \$1 million

The representations and warranties of each party set forth in the Merger Agreement have been made solely for the benefit of the other parties to the Merger Agreement, and these representations and warranties should not be relied on by any other person. In addition, these representations and warranties may have been qualified by disclosures made to the other parties in connection with the Merger Agreement, are subject to a materiality standard that may differ from what may be viewed as material by investors, and were made only as of the date of the Merger Agreement or such other date as is specified in the Merger Agreement. Accordingly, investors should not rely on the representations and warranties as characterizations of the actual state of facts or for any other purpose at the time they were made or otherwise.

The foregoing description of the Merger Agreement does not purport to be complete and is qualified in its entirety by reference to the Merger Agreement, which is filed as Exhibit 2.1 to this Form 8-K, and is incorporated by reference herein.

On September 21, 2007, the Company and Rockhill issued a press release announcing that they had entered into the Merger Agreement. A copy of the press release is filed as Exhibit 99.1 to this Form 8-K and is incorporated herein by reference.

Amendment to the Equity Plans and the ESPP

On September 20, 2007, the Company's Board of Directors approved amendments to the Company's 1994 Stock Plan and the 2005 Stock Plan (collectively, the "Equity Plans") to give effect to certain provisions of the Merger Agreement.

The Board of Directors amended the Equity Plans to provide that all options outstanding under the Equity Plans will automatically be accelerated and will become fully vested and exercisable immediately prior to the effective time of the Merger. The Board of Directors also amended the Equity Plans to provide that all outstanding options under the Equity Plans at the effective time of the Merger will be cancelled and the holders of these options will receive a cash payment, in lieu of any interest in the Company's common stock or the common stock of the other entity, equal to the number of shares underlying the options multiplied by the excess, if any, of the merger consideration over the exercise price. Under the Equity Plans, the Company may also deduct from this cash payment an amount required to be withheld by the Company with respect to income taxes with respect to the option holders. Finally, the Board of Directors amended the Equity Plans to provide that, at the effective time of such merger, the Equity Plans will be terminated and shares of the Company's common stock no longer will be reserved for issuance under the Equity Plans. The foregoing description of the amendments to the Equity Plans does not purport to be complete and is qualified in its entirety by reference to the form of amendments to the Equity Plans attached hereto as Exhibit 10.4 and 10.5 and incorporated herein by reference.

Additionally on September 20, 2007, the Board of Directors approved amendments to the 1995 Employee Stock Purchase Plan (the "ESPP") to provide that at the effective time of the Merger, options then outstanding under the ESPP will be cancelled and exchanged for a cash payment equal to the whole number of shares the participant would be otherwise entitled to purchase under the ESPP multiplied by the excess of \$12.45 over the exercise price as determined under the ESPP.

Additionally, the current offering period of the ESPP will terminate on the earlier of (i) December 31, 2007 and (ii) immediately prior to the effective time of the Merger. The foregoing description of the amendment to the ESPP does not purport to be complete and is qualified in its entirety by reference to the amendment to the ESPP attached hereto as Exhibit 10.6 and incorporated herein by reference.

The full text of each of the Equity Plans and the ESPP immediately prior to the amendments of September 20, 2007 are incorporated herein by reference as described in Exhibits 10.1 through 10.3 of Item 9.01 to this Form 8-K.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Officers; Compensatory Arrangements of Certain Officers.

Employment Agreements

Among the conditions to consummation of the Merger is that the following RTW executives execute and be able to perform under employment agreements with Rockhill following the Merger: Jeffrey B. Murphy, Alfred L. LaTendresse, Keith D. Krueger, David M. Dietz and Patricia M. Sheveland. The terms of the employment arrangements have been reviewed and approved by the Company's Board of Directors in connection with the Board's approval of the Merger Agreement. The employment agreements provide for a base salary, participation in a bonus program, restricted stock awards, directors and officers' liability insurance, and participation in Rockhill's benefits programs. Mr. Murphy will also be eligible for \$2 million of term life insurance.

The following chart summarizes the basic terms of the employment arrangements.

Summary of Employment Arrangements Following the Merger

Executive	Title	Current RTW Salary (\$)	New Salary Following Merger (\$)	Restricted Share Award (#)	Bonus Program Eligibility	Term of Agreement	Non- Competition Covenant
Jeffrey B.	President, Chief Executive Officer	350,000	350,000	1,000	Yes	2 years	2 years
Murphy							
Alfred L.	Executive Vice President, Chief Financial Officer	220,000	220,000	400	Yes	1 year	1 year
LaTendresse							
Keith D.	Chief Operating Officer	205,000	205,000	600	Yes	1 year	1 year
Krueger							
David M.	VP-Business Development	177,000	177,000	600	Yes	1 year	1 year
Dietz							
Patricia M.	VP-Product Development, Quality & Compliance	177,000	177,000	400	Yes	1 year	1 year
Sheveland							

Under the employment agreements, if an executive's employment terminates due to death or disability, the executive is entitled to payment of his or her base salary, and any earned or accrued benefits under the restricted stock or bonus programs, and in the case of disability, the executive and his or dependents may continue to participate in medical, dental and life insurance plans for one year. If the executive is terminated for cause, the executive receives his or her base salary through the last day of employment and payment of any earned or accrued benefits under the restricted stock or bonus programs. If the executive is terminated without cause or the executive terminates employment for good reason, then the executive is entitled to: (i) base salary for one year following termination in the case of Mr. Murphy and the greater of six months or the two weeks for each year of service for the other officers; (ii) continued participation in medical, dental and life insurance plans for one year for the executive and dependents; (iii) payment of any earned or accrued benefits under the restricted stock or bonus programs. Cause is defined as the executive's: (i) failure to perform his or her duties, (ii) engaged in malfeasance, fraud or dishonesty, (iii) violation of company policies and practices that has a material adverse effect, (iv) material breach of the employment agreement, (v) conviction, guilty or no contest plea to a crime involving moral turpitude or a felony. Good reason means: (i) a material change in the executives duties; (ii) relocation out of Minneapolis; (iii) material breach by RTW of the employment agreement; or (iv) a change in control at Rockhill. The executives also agree that in the event of termination of employment for any reason, the executive cannot compete with Rockhill for a period of one year, except for Mr. Murphy whose non-competition covenant is for two years.

Bonus Plan

The executives are also eligible to participate in a bonus program. The bonus program is calculated and paid based on a percentage of underwriting profit. Underwriting profit is based on net earned premium less estimated ultimate losses and loss adjustment expense less other operating expenses. Bonuses are calculated at 7.5% of underwriting profit with a minimum of \$750,000 and a maximum of \$3 million in any calendar year to be allocated among all five of the RTW executives and other executives of Rockhill in amounts determined by Rockhill's chief executive officer. Bonus payments are paid out over three or six year periods and bonus plan eligibility is determined by Rockhill's President and Chief Executive Officer. For example, if the Company-wide underwriting profit for a given year were \$20,000,000, then a bonus of \$1,500,000 would be eligible for distribution to all plan participants.

Restricted Stock Plan

Each of the executives will also receive an award of restricted shares under Rockhill's 2005 Restricted Stock Program. The executives will each receive an award in the amount described under each executive's name in the chart above. The restricted shares vest after five years only if the executive remains employed with Rockhill, and Rockhill achieves performance goals related to either return on average equity or internal rate of return targets in excess of 13% and 25% respectively, in the performance period from 2008 to 2012. In the event of a change in control of Rockhill, the restricted shares vest in full. Each share of restricted stock is valued at approximately \$1,000 per share and the aggregate number of shares issuable to RTW if the targets are achieved in the performance period is 3,000 shares, representing just over 2% of the total shares outstanding of Rockhill.

The foregoing descriptions of the employment agreements and restricted stock plan do not purport to be complete and are qualified in their entirety by reference to the form of employment agreement and form of Restricted Stock Plan Award Agreement attached hereto as Exhibits 10.7 and 10.8 and incorporated herein by reference.

Information Regarding Former Executive

As previously disclosed by the Company, on June 8, 2007, the employment of Mr. Thomas J. Byers, Executive Vice President – Sales and Marketing, as an officer and employee of RTW, Inc. terminated. In connection with the termination his employment, the Company agreed to continue Mr. Byers' salary for six months and also agreed that if a transaction involving the Company occurred prior to December 31, 2007, then Mr. Byers would be entitled to an additional payment equal to the difference between his option exercise prices and the transaction price. As a result of the Merger, Mr Byers will receive a payment of \$14,325 representing the difference between the exercise price of his options and the merger consideration.

Item 8.01 Other Events

Attached as Exhibit 99.2 is a Question and Answer Fact Sheet regarding the proposed acquisition of the Company by Rockhill that was used by the Company's management to respond to questions to employees regarding the Merger.

Important Merger Information

In connection with the proposed Merger, the Company will file a proxy statement with the U.S. Securities and Exchange Commission, or SEC. Shareholders and investors are advised to read the proxy statement when it becomes available because it will contain important information about the Merger and the Company. Shareholders and investors may obtain a free copy of the proxy statement (when available) and other documents filed by the Company with the SEC at the SEC's web site at www.sec.gov. Free copies of the proxy statement, once available, and the Company's other filings with the SEC, may also be obtained from the Company at www.rtwi.com by clicking on the "Investors" tab and then following the link at "Financial Reports." Free copies of the Company's filings may be obtained by directing a written request to RTW, Inc., 8500 Normandale Lakes Blvd., Bloomington, Minnesota 55437, Attention: Alfred L. LaTendresse, or by telephone at (952) 893-0403.

Participants in the Solicitation

The Company and its directors, executive officers and other members of its management may be deemed to be soliciting proxies from the Company's shareholders in favor of the Merger. Investors and shareholders may obtain more detailed information regarding the direct and indirect interests in the Merger of persons who may, under the rules of the SEC, be considered participants in the solicitation of the Company's shareholders in connection with the merger by reading the preliminary and definitive proxy statements regarding the Merger, which will be filed with the SEC. Information about the Company's directors and executive officers may be found in the Company's definitive proxy statement filed with the SEC on April 27, 2007. These documents will be available free of charge once available at the SEC's web site at www.sec.gov or by directing a request to the Company as described above.

Item 9.01 Financial Statements And Exhibits

Exhibit No. Description

- 2.1 Agreement and Plan of Merger dated as of September 20, 2007 among Rockhill Holding Company, Rockhill Acquisition Corporation and RTW, Inc.
- Amended RTW, Inc. 1995 Employee Stock Purchase Plan (incorporated by reference to Exhibit 4.2 to the Company's Registration Statement on Form S-8 (Reg. No. 333-127107)(August 2, 2005).
- Amended RTW, Inc. 1994 Stock Plan (incorporated by reference to Exhibit 4.1 to the Company's Registration Statement on Form S-8 (Reg. No. 333-81408)(January 25, 2002).
- RTW, Inc. 2005 Stock Plan (incorporated by reference to Exhibit 4.1 to the Company's Registration Statement on Form S-8 (Reg. No. 333-127107)(June 5, 1997127107)(August 2, 2005).
- Amendment to and termination of RTW, Inc. 1995 Employee Stock Purchase Plan dated September 20, 2007.
- 10.5 Amendment to and termination of RTW, Inc. 1994 Stock Plan dated September 20, 2007.
- 10.6 Amendment to and termination of RTW, Inc. 2005 Stock Plan dated September 20, 2007.
- 10.7 Form of Employment Agreement.
- 10.8 Form of Restricted Stock Grant Agreement.*
- 99.1 Press Release issued on September 21, 2007 by RTW, Inc. and Rockhill Holding Company.
- 99.2 Question and Answer Fact Sheet dated September 26, 2007 regarding Proposed Acquisition of RTW, Inc. by Rockhill Holding Company.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

RTW, INC.

By: /s/ Jeffrey B. Murphy Jeffrey B. Murphy President and Chief Executive Officer

Date: September 26, 2007

^{*} To be filed by amendment.

<DOCUMENT>
<TYPE> EX-2.1
<FILENAME> c18876exv2w1.htm
<DESCRIPTION> Agreement and Plan of Merger
<TEXT>

AGREEMENT AND PLAN OF MERGER DATED AS OF SEPTEMBER 20, 2007 AMONG ROCKHILL HOLDING COMPANY, ROCKHILL ACQUISITION CORPORATION AND RTW, INC.

TABLE OF CONTENTS

ARTICLE 1. DEFINITIONS	Page 1
ARTICLE 2. TERMS OF MERGER	6
2.1. Effect of Merger and Surviving Corporation	6
2.2. Stock of Company	7
2.3. Company Stock Options	7
2.4. Effect on Merger Sub Stock	7
2.5. Exchange Procedures	8
2.6. Adjustments	9
2.7. Directors of Surviving Corporation	9
2.8. Executive Officers of Surviving Corporation	9
2.9. No Further Ownership Rights in Stock	9
2.10. Articles of Incorporation and Bylaws	9
ARTICLE 3. THE CLOSING	9
3.1. Closing Date	9
3.2. Articles of Merger	10
3.3. Further Assurances	10
ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF COMPANY	10
4.1. Incorporation, Standing and Power	10
4.2. Capitalization.	10
4.3. Subsidiaries	11
4.4. Financial Statements	11
4.5. Reports and Filings.	12
4.6. Authority of Company	13
4.7. Insurance	13
4.8. Personal Property	14
4.9. Real Estate	14
4.10. Litigation	14
4.11. Taxes	14
4.12. Compliance with Charter Provisions and Laws and Regulations	17
4.13. Employees	18
4.14. Brokers and Finders	18
4.15. Scheduled Contracts	18
4.16. Performance of Obligations	19
4.17. Certain Material Changes	19
4.18. Licenses and Permits	20
4.19. Undisclosed Liabilities	20
4.20. Employee Benefit Plans.	20
4.21. Corporate Records	22
4.22. Accounting Records and Internal Controls	22

4.23. Vote Required	<u>Pag</u>
4.24. Disclosure Documents and Applications	22
4.25. Intellectual Property	23
4.26. State Takeover Laws	24
4.27. Opinion of KBW	24
4.28. Insurance Matters	24
4.29. Restrictions on Business Activities	26
4.30. No Additional Representations	26
ARTICLE 5. REPRESENTATIONS AND WARRANTIES OF PARENT	27
5.1. Incorporation, Standing and Power	27
5.2. Authority	27
5.3. Financing	28
5.4. Litigation	28
5.5. Ownership of Merger Sub	28
5.6. Facts Affecting Regulatory Approvals	28 28
5.7. Accuracy of Information Furnished for Company Proxy Statement	20
ARTICLE 6. COVENANTS OF COMPANY PENDING EFFECTIVE TIME OF THE MERGER	28
6.1. Limitation on Conduct Prior to Effective Time of the Merger	28
6.2. Affirmative Conduct Prior to Effective Time of the Merger	32
6.3. Access to Information	33
6.4. Filings	33
6.5. Notices; Reports	34
6.6. Company Shareholders' Meeting	34
6.7. Proxy Statement	34
ARTICLE 7. COVENANTS OF PARENT AND MERGER SUB	35
7.1. Limitation on Conduct Prior to Effective Time of the Merger	35
7.2. Applications	35
7.3. Notices; Reports	35
7.4. Indemnification and Directors' and Officers' Insurance	35
ARTICLE 8. ADDITIONAL COVENANTS	36
8.1. HSR Matters	36
8.2. Insurance Approvals.	37
8.3. Commercially Reasonable Efforts	37
8.4. Public Announcements	37
ARTICLE 9. CONDITIONS PRECEDENT TO THE MERGER	38
9.1. Shareholder Approval	38
9.2. Insurance Approvals	38
9.3. No Judgments or Orders	38

9.4. HSR Approvals 9.5. Employment Agreements	Pag 38 38
9.6. Claim & Claim Settlement Expense9.7. Minnesota Workers' Compensation Assigned Risk Plan	38 38
ARTICLE 10. CONDITIONS PRECEDENT TO THE OBLIGATIONS OF COMPANY	39
10.1. Representations and Warranties; Performance of Covenants10.2. Officers' Certificate10.3. Employee Benefit Plans	39 39 39
ARTICLE 11. CONDITIONS PRECEDENT TO OBLIGATIONS OF PARENT AND MERGER SUB	39
 11.1. Representations and Warranties; Performance of Covenants 11.2. Authorization of Merger 11.3. Officers' Certificate 11.4. Company Dissenting Shares 11.5. Employee Benefit Plans 11.6. Transaction Related Expenses 11.7. No Material Adverse Effect 	39 40 40 40 40 40
ARTICLE 12. EMPLOYEE BENEFITS	40
12.1. Employee Benefits12.2. Company Stock Options and the Company Stock Option Plans	40 42
ARTICLE 13. TERMINATION	43
13.1. Termination 13.2. Effect of Termination	43 44
ARTICLE 14. MISCELLANEOUS	45
 14.1. Expenses 14.2. Notices 14.3. Assignment 14.4. Counterparts 14.5. Effect of Representations and Warranties 14.6. Third Parties 14.7. Integration 14.8. Knowledge 14.9. Governing Law 	45 46 46 46 46 46 46 46
14.10. Captions 14.11. Severability 14.12. Waiver and Modification; Amendment	47 47 47

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER ("<u>Agreement</u>") is made and entered into as of the 20th day of September, 2007, by and among Rockhill Holding Company, a Delaware corporation ("<u>Parent</u>"), Rockhill Acquisition Corporation, a Minnesota corporation and a wholly-owned subsidiary of Parent ("<u>Merger Sub</u>"), and RTW, INC., a Minnesota corporation ("<u>Company</u>").

WHEREAS, each of Parent, Merger Sub and Company desires to enter in to a transaction whereby Merger Sub will merge with and into Company (the "Merger"), with Company being the surviving corporation, upon the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the Board of Directors of the Company has approved the Merger, the Agreement and the transactions contemplated therein in accordance with the provisions of the Minnesota Business Corporations Act (the "MBCA") and determined the Merger is advisable and in the best interests of its shareholders upon the terms and conditions set forth herein and in accordance with the MBCA (Company, following the effectiveness of the Merger, being hereinafter sometimes referred to as the "Surviving Corporation"); and

WHEREAS, the Board of Directors of Parent, Merger Sub and Company each have approved this Agreement and the Merger pursuant to which Merger Sub will merge with and into Company and each outstanding share of Company common stock, no par value per share ("Company Stock"), excluding any Company Dissenting Shares (as defined below), will be converted into the right to receive the Merger Consideration (as defined in Section 2.2(b)) upon the terms and subject to the conditions set forth herein.

NOW, THEREFORE, on the basis of the foregoing recitals and in consideration of the respective covenants, agreements, representations and warranties contained herein, the parties hereto agree as follows:

ARTICLE 1.

DEFINITIONS

Except as otherwise expressly provided for in this Agreement, or unless the context otherwise requires, as used throughout this Agreement the following terms shall have the respective meanings specified below:

"ACIC" means American Compensation Insurance Company, a wholly owned subsidiary of Company.

"Affiliate" of, or a Person "Affiliated" with, a specific Person(s) is a Person that directly or indirectly, through one or more intermediaries, controls, or is controlled by, or is under common control with, the Person(s) specified.

"Affiliated Group" means, with respect to any entity, a group of entities required or permitted to file consolidated, combined or unitary Tax Returns (as defined herein).

- "Articles of Merger" has the meaning set forth in Section 3.2.
- "BCIC" means Bloomington Compensation Insurance Company, a wholly owned subsidiary of ACIC.
- "Benefit Arrangements" has the meaning set forth in Section 4.20(b).
- "Book Entry Shares" has the meaning set forth in Section 2.5(b).
- "Business Day" means any day other than a Saturday, Sunday or other day on which banks in New York and Minnesota are required or authorized by law to be closed.
 - "Certificates" has the meaning set forth in Section 2.5(b).
- "Closing" means the consummation of the Merger provided for in Article 2 of this Agreement on the Closing Date (as defined herein) at the offices of Rockhill Holding Company, 700 West 47th Street, Suite 350, Kansas City, Missouri 64112, or at such other place as the parties may agree upon.
- "Closing Date" means the date that is no later than the second business day following the day on which the last of the conditions specified in Articles 9, 10 and 11 (excluding, for purposes of this definition, conditions that, by their terms, are to be satisfied on the Closing Date) have been fulfilled or waived (if permissible) or such other date as the parties may agree upon.
 - "Code" means the Internal Revenue Code of 1986, as amended.
 - "Company 401(k) Plan" means the RTW, Inc. KSOP Plan.
- "Company Disclosure Letter" means that letter designated as such that has been delivered by Company to Parent prior to the execution and delivery of this Agreement.
 - "Company Dissenting Shares" has the meaning set forth in Section 2.2(b).
 - "Company Option List" has the meaning set forth in Section 4.2(a).
 - "Company Patents" has the meaning set forth in Section 4.25(b).
 - "Company Property" has the meaning set forth in Section 4.12(b).
 - "Company Registered IP" has the meaning set forth in Section 4.25(b).
 - "Company Registered Marks" has the meaning set forth in Section 4.5(b).
 - "Company SEC Documents" has the meaning set forth in Section 4.5(a).
 - "Company Shareholders' Meeting" means the meeting of Company's shareholders referred to in Section 6.6.
 - "Company Stock" has the meaning set forth in the second recital of this Agreement.

"Company Stock Option Plans" means, collectively, the 1994 Stock Plan, as amended, the 2005 Stock Plan, as amended and the 1995 Employee Stock Purchase Plan and Trust.

"Company Stock Option" means any option or right to acquire Company Stock, or stock appreciation right payable in cash issued pursuant to Company Stock Option Plans.

"Company Supplied Information" has the meaning set forth in Section 4.24.

"Competing Transaction" has the meaning set forth in Section 6.1(m).

"Confidentiality Agreement" means that certain Confidentiality Agreement dated June 28, 2007 by and between Rockhill Insurance Company and Company.

"Copyrights" has the meaning set forth in Section 4.25(a).

"E&Y" means Ernst & Young LLP, Company's independent public accountants.

"Effective Time of the Merger" means the date upon which the Articles of Merger is filed with the Secretary of State of the State of Minnesota, or at such time thereafter as shall be agreed to by the parties and specified in the Articles of Merger.

"Employee Plans" has the meaning set forth in Section 4.20(a).

"Encumbrance" shall mean any option, pledge, security interest, lien, charge, encumbrance or restriction (whether on voting or disposition or otherwise), whether imposed by agreement, understanding, law or otherwise.

"Environmental Regulations" has the meaning set forth in Section 4.12(b).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliates" has the meaning set forth in Section 4.20(a).

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Exchange Agent" means a bank or trust company that the parties will designate.

"Exchange Fund" has the meaning set forth in Section 2.5(a) hereof.

"Executive Employment Agreement" means the existing employment agreement between the Company and its President and Chief Executive Officer.

"Expenses" has the meaning set forth in Section 14.1 hereof.

"Financial Statements of Company" means the financial statements of Company consisting of the balance sheets as of December 31, 2004, 2005 and 2006, the related statements of income, shareholders' equity and cash flows for the years then ended and the related notes thereto and related opinions of E&Y thereon for the years then ended, and any interim financial statements since December 31, 2006, filed as part of the SEC Documents.

"GAAP" means United States generally accepted accounting principles consistently applied during the periods involved.

"Governmental Entity" means any court, tribunal or judicial or arbitral body in any jurisdiction or any United States federal, state, municipal or local or any foreign or other governmental, regulatory or administrative authority, agency or instrumentality.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Hazardous Materials" has the meaning set forth in Section 4.12(b).

"Indemnified Liabilities" has the meaning set forth in Section 7.4(a).

"Indemnified Parties" has the meaning set forth in Section 7.4(a).

"Intellectual Property" has the meaning set forth in Section 4.25(a).

"IRS" means the Internal Revenue Service.

"KBW" means Keefe, Bruyette & Woods, Inc. the Company's investment banker.

"KBW Agreement" means the letter agreement dated January 19, 2007 between Company and KBW.

"Laws" means all federal, state, local or foreign or provincial laws, statutes, ordinances, rules, regulations, judgments, orders, injunctions, decrees or agency requirements of or undertaking to or agreement with any Governmental Entity, including common law.

"MBCA" has the meaning set forth in the second recital of this Agreement.

"Marks" has the meaning set forth in Section 4.25(a)

"Material Adverse Effect" means any circumstance, change in or effect on Company, ACIC, BCIC or the Surviving Corporation (1) that is, or would reasonably be expected to be, materially adverse to the condition (financial or otherwise), business, properties, assets, liabilities, prospects, or results of operations of Company, ACIC, BCIC or the Surviving Corporation, taken as a whole, or (2) that materially impairs or would reasonably be expected to materially impair the ability of Company to timely perform its obligations under this Agreement or to consummate the transactions contemplated hereby; provided, however, that in determining whether a Material Adverse Effect has occurred there shall be excluded the effect of: (i) any change in GAAP, Statutory Accounting Policies, or regulatory accounting requirements applicable to the insurance industry generally but not such changes that have a disproportionate effect on the Company, (ii) any general social, political, economic, environmental or natural condition, change, effect, event or occurrence the effects of which are not specific or unique to Company, including changes in prevailing interest rates, currency exchange rates or general global economic or global market conditions, (iii) any failure by the Company to meet any published projections, forecasts, or predictions of revenue or earnings for any period, (iv) any

loss or threatened loss of business from any agents, brokers, or customers of the Company or its Subsidiaries caused by the announcement or the pendency of the transactions contemplated by this Agreement, (v) any action or omission by Company expressly permitted under this Agreement including the public announcement of the transactions contemplated by this Agreement, (vi) any expenses incurred in connection with this Agreement or the transactions contemplated hereby, and (vii) the payment of any amounts due to, or the provision of any other benefits to, any officers or employees under contracts or plans in existence on the date of this Agreement and disclosed in the Company Disclosure Letter.

- "Merger" has the meaning set forth in the first recital of this Agreement.
- "Merger Consideration" has the meaning set forth in Section 2.2(a).
- "Other Incentive Plans" has the meaning set forth in Section 12.1(c)(ii).
- "Patents" has the meaning set forth in Section 4.25(a).
- "Person" means any individual, corporation, association, partnership, limited liability company, trust, joint venture, other entity, unincorporated organization, government or governmental department or agency.
- "Proxy Statement" means the Proxy Statement, together with any supplements thereto, that is used to solicit proxies for the Company Shareholders' Meeting in connection with the Merger.
 - "Representatives" has the meaning set forth in Section 6.1(m).
 - "Scheduled Contract" has the meaning set forth in Section 4.15.
 - "SEC" means the Securities and Exchange Commission.
 - "Securities Act" means the Securities Act of 1933, as amended.
- "Statutory Accounting Policies" means the statutory accounting practices prescribed or permitted by the Minnesota Department of Commerce for determining and reporting the financial condition and results of operations of insurance companies and determining the solvency of insurance companies under Minnesota law as set forth in the National Association of Insurance Commissioners' Accounting Practices and Procedures Manual, version effective January 1, 2006 as adopted by the Minnesota Department of Commerce.
- "Subsidiary" of a Person means any corporation, partnership, limited liability company or other business entity of which more than 25% of the voting power is owned or controlled by such Person.
 - "Superior Proposal" has the meaning set forth in Section 6.1(m).
 - "Surviving Corporation" has the meaning set forth in the first recital of this Agreement.

"Tank" has the meaning set forth in Section 4.12(b).

"Tax" or "Taxes" means (i) any and all federal, state, local or foreign taxes, charges, premium taxes, fees, imposts, levies or other assessments, including, without limitation, all net income, gross receipts, capital, sales, use, ad valorem, value added, transfer, franchise, profits, inventory, capital stock, license, withholding, payroll, employment, social security, unemployment, excise, severance, stamp, occupation, property, corporation and estimated taxes, custom duties, fees, assessments and charges of any kind whatsoever; (ii) all interest, penalties, fines, additions to tax or additional amounts imposed by any taxing authority in connection with any item described in clause (i); and (iii) any transferred liability in respect of any items described in clauses (i) or (ii).

"Tax Returns" means all material returns, sales and use returns, declarations, reports, information returns, statements, elections, disclosures and schedules required to be filed in respect of any Taxes (including any attachments thereto or amendments thereof).

"Termination Fee" has the meaning set forth in Section 13.2(b).

"Trade Secrets" has the meaning set forth in Section 4.25(a).

"Transaction-Related Expenses" means all expenses incurred by Company in connection with or related to the authorization, preparation and execution of this Agreement, the solicitation of shareholder approvals and all other matters related to the closing of the transactions contemplated hereby, including without limitation, the HSR Act filing fees, fees of Representatives, including KBW, and all related other fees and expenses or agents, representatives, counsel and accountants employed by Company or its Subsidiaries or Affiliates from the date of this Agreement. "Transaction-Related Expenses" does not include: (i) payments to any officer, director or employee permitted under Article 12 hereof; (ii) legal and other expenses including additional investment banking and other fees related to Company's response in the event of a third party's proposal that might be considered or lead to a Competing Transaction or Superior Proposal; or (iii) legal or other costs and expenses related to Company's response to any claim, action, suit, proceeding or investigation of any Governmental Entity or any third party based in whole or in part, or arising from in whole or in part, this Agreement, or regulatory or other approvals or investigations or inquiries related to this Agreement and the transactions contemplated by this Agreement.

ARTICLE 2.

TERMS OF MERGER

2.1. Effect of Merger and Surviving Corporation. At the Effective Time of the Merger, Merger Sub will be merged with and into Company pursuant to the terms, conditions and provisions of this Agreement and in accordance with the applicable provisions of the MBCA, and the separate corporate existence of Merger Sub shall cease. The Merger will have the effects set forth in the MBCA.

- 2.2. Stock of Company. Subject to Section 2.6, each share of Company Stock issued and outstanding immediately prior to the Effective Time of the Merger shall, without any further action on the part of Company or the holders of such shares, be treated on the basis set forth in this Section 2.2.
- (a) <u>Cancellation of Parent Owned Stock</u>. At the Effective Time of the Merger, any share of the Company Stock held by the Company as treasury stock, held by ACIC or BCIC, or owned by Parent or Merger Sub shall be automatically cancelled and retired and shall cease to exist and no Merger Consideration shall be delivered therefore.
- (b) <u>Conversion of Company Stock</u>. At the Effective Time of the Merger, each issued and outstanding share of Company Stock (other than shares that will converted into the right to receive the consideration determined under Section 12.2(a), shares that will be cancelled in accordance with Section 2.2(a) and any Company Dissenting Shares) shall be automatically canceled and cease to be an issued and outstanding share of Company Stock and be converted into the right to receive per share consideration (the "<u>Merger Consideration</u>") in cash in the amount of \$12.45.
- (c) Company Dissenting Shares. Notwithstanding anything in this Agreement to the contrary, any shares of Company Stock that are issued and outstanding as of the Effective Time of the Merger and that are held by a shareholder of Company who has properly exercised such holder's dissenters' rights under the MBCA (the "Company Dissenting Shares") shall not be converted into the right to receive the Merger Consideration unless and until such holder shall have failed to perfect, or shall have effectively withdrawn or lost, such holder's right to dissent from the Merger under the MBCA, but will be converted to the right receive such consideration as may be determined to be due with respect to such Company Dissenting Shares pursuant to and subject to the requirements of the MBCA. If any such holder shall have so failed to perfect or have effectively withdrawn or lost such right at the Effective Time of the Merger, each share of such holder's Company Stock shall thereupon be deemed to have been converted into and to have become, as of the Effective Time of the Merger, the right to receive, without any interest thereon, the Merger Consideration. Company shall give Parent (i) prompt notice of any notice or demands for appraisal or payment for shares of Company Stock received by Company and (ii) the opportunity to participate in all negotiations and proceedings with respect to any such demands or notices. Company shall not, without the prior written consent of Parent, or as required by MBCA, make any payment with respect to, or settle, offer to settle or otherwise negotiate, any such demands.
- 2.3. <u>Company Stock Options</u>. Each Company Stock Option outstanding as of the Effective Time of the Merger shall be treated in accordance with Section 12.2.
- 2.4. <u>Effect on Merger Sub Stock</u>. At the Effective Time of the Merger, each issued and outstanding share of capital stock of Merger Sub shall be converted into and become one fully paid and nonassessable share of common stock of the Surviving Corporation.

2.5. Exchange Procedures.

- (a) At the Effective Time of the Merger, Parent shall deposit with the Exchange Agent for the benefit of the holders of shares of Company Stock outstanding immediately prior to the Effective Time of the Merger, for exchange in accordance with this Section 2.5 through the Exchange Agent, cash in the amount of the Merger Consideration payable to such holders of Company Stock pursuant to Section 2.2 in exchange for their shares of Company Stock (collectively, the "Exchange Fund").
- (b) Parent shall direct the Exchange Agent to mail, promptly after the Effective Time of the Merger, to each holder of record of shares of Company Stock that are represented by (x) a certificate or certificates that immediately prior to the Effective Time of the Merger represented outstanding shares of Company Stock (the "Certificates") or (y) an entry to that effect in the shareholder records maintained on behalf of Company by the Company stock transfer agent (the "Book Entry Shares"), whose shares were converted into the right to receive the Merger Consideration pursuant to Section 2.2 hereof, (i) a letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates (if any) shall pass, only upon delivery of the Certificates to the Exchange Agent and shall be in such form and have such other provisions as Parent and Company may reasonably specify), and (ii) instructions for use in effecting the surrender of the Certificates or authorizing transfer and cancellation of Book Entry Shares in exchange for the Merger Consideration. Upon surrender of a Certificate for cancellation to the Exchange Agent or to such other agent or agents as may be appointed by Parent, or authorizing transfer of Book Entry Shares, together with such letter of transmittal, duly executed, the holder of such shares of Company stock shall be entitled to receive in exchange therefore the amount of the Merger Consideration that such holder has the right to receive pursuant to Section 2.2 hereof, and any Certificate so surrendered shall forthwith be canceled. Until surrendered as contemplated by this Section 2.5, each Certificate and any Book Entry Shares shall be deemed at any time after the Effective Time of the Merger to represent only the right to receive upon such surrender the Merger Consideration to be paid in consideration therefore upon surrender of such Certificate or transfer of the Book Entry Shares, as the case may be, as contemplated by this Section 2.5. Notwithstanding anything to the contrary set forth herein, if any holder of shares of Company Stock that are not Book Entry Shares should be unable to surrender the Certificates for such shares, because they have been lost or destroyed, such holder shall, if required by Parent or Exchange Agent, deliver in lieu thereof a bond in form and substance and with surety reasonably satisfactory to Parent and shall be entitled to receive the Merger Consideration to be paid in consideration therefore in accordance with Section 2.2 hereof.
- (c) If, after the Effective Time of the Merger, Certificates or Book Entry Shares are presented to Parent for any reason, they shall be canceled and exchanged as provided in this Agreement.
- (d) Any portion of the Exchange Fund that remains undistributed to the shareholders of Company following the passage of twelve months after the Effective Time of the Merger shall be delivered to the Surviving Corporation, upon demand, and any shareholders of Company who have not theretofore complied with this Section 2.5 shall thereafter look only to

the Surviving Corporation and Parent for payment of their claim for the Merger Consideration payable in consideration for any Certificate or transfer of any Book Entry Shares.

- (e) Except as otherwise required by Law, neither Parent nor the Surviving Corporation shall be liable to any holder of shares of Company Stock for such cash from the Exchange Fund delivered to a public official pursuant to any applicable abandoned property, escheat or similar law.
- 2.6. Adjustments. If after the date hereof and on or prior to the Effective Time of the Merger but subject to the prior written consent of Parent, the outstanding shares of Company Stock shall be changed into a different number of shares by reason of any reclassification, recapitalization or combination, stock split, reverse stock split, stock dividend or rights issued in respect of such stock, or any similar event shall occur, the Merger Consideration shall be adjusted accordingly to provide to the holders of Company Stock the same economic effect as contemplated by this Agreement prior to such event.
- 2.7. <u>Directors of Surviving Corporation</u>. At the Effective Time of the Merger, the Board of Directors of the Surviving Corporation shall be comprised of the persons serving as directors of Merger Sub immediately prior to the Effective Time of the Merger. Such persons shall serve until the earlier of their resignation or removal or until their respective successors are duly elected and qualified.
- 2.8. Executive Officers of Surviving Corporation. At the Effective Time of the Merger, the executive officers of the Surviving Corporation shall be comprised of the persons serving as executive officers of Merger Sub immediately prior to the Effective Time of the Merger. Such persons shall serve until the earlier of their resignation or termination.
- 2.9. No Further Ownership Rights in Stock. All Merger Consideration delivered upon the surrender for exchange of shares of Company Stock in accordance with the terms hereof shall be deemed to have been delivered in full satisfaction of all rights pertaining to ownership of such shares of stock. At and after the Effective Time of the Merger, there shall be no further registration of transfers on the stock transfer books of the Surviving Corporation of the shares of Company Stock that were outstanding immediately prior to the Effective Time of the Merger, and upon delivery of the Merger Consideration upon surrender for exchange of Company Stock, each such share of Company Stock shall be canceled.
- 2.10. <u>Articles of Incorporation and Bylaws</u>. The articles of incorporation of Merger Sub as in effect immediately prior to the Effective Time of the Merger shall be the articles of incorporation of the Surviving Corporation. The bylaws of Merger Sub as in effect immediately prior to the Effective Time of the Merger shall be the bylaws of the Surviving Corporation.

ARTICLE 3. THE CLOSING

3.1. Closing Date. The Closing shall take place on the Closing Date.

- 3.2. <u>Articles of Merger</u>. Subject to the provisions of this Agreement, a certificate of merger (the "<u>Articles of Merger</u>") shall be duly prepared, executed by the Surviving Corporation and thereafter delivered to the Secretary of State of the State of Minnesota for filing, as provided in the MBCA, on the Closing Date.
- 3.3. <u>Further Assurances</u>. At the Closing, the parties hereto shall deliver, or cause to be delivered, such documents or certificates as may be necessary in the reasonable opinion of counsel for any of the parties, to effectuate the transactions contemplated by this Agreement.

ARTICLE 4.

REPRESENTATIONS AND WARRANTIES OF COMPANY

The following representations and warranties by Company to Parent and Merger Sub are qualified by the Company Disclosure Letter. The Company Disclosure Letter shall refer to the representation or warranty to which exceptions or matters disclosed therein relate except that an exception or matter disclosed with respect to one representation or warranty shall also be deemed disclosed with respect to each other warranty or representation to which the exception or matter reasonably relates. The inclusion of any item in the Company Disclosure Letter shall not be deemed an admission that such item is a material fact, event or circumstance or that such item has or had, or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

4.1. <u>Incorporation, Standing and Power</u>. Each of Company, ACIC and BCIC has been duly organized, are validly existing and in good standing as a corporation under the laws of the State of Minnesota. Each of the Company, ACIC and BCIC have all requisite corporate power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted. Each of the Company, ACIC and BCIC is duly qualified to do business in each jurisdiction where the character of its properties owned or held under lease or the nature of its activities makes such qualification necessary. Company has delivered to Parent true and correct copies of the Company's articles of incorporation and bylaws, ACIC's articles of incorporation and bylaws, and BCIC's articles of incorporation and bylaws, respectively, as currently in effect.

4.2. Capitalization.

(a) As of the close of business on September 19, 2007, the authorized capital stock of Company consists of 12,500,000 shares of Company Stock, of which 5,174,845 shares of common stock are outstanding, and 4,750,000 shares of preferred stock, 250,000 shares have been authorized and none of which are outstanding. All of the outstanding shares of Company Stock are validly issued, fully paid and nonassessable. Except for Company Stock Options covering 580,381 shares of Company Stock granted pursuant to the Company Stock Option Plans, there are no outstanding options, warrants or other rights in or with respect to the unissued shares of preferred stock, Company Stock nor any securities convertible into such stock, and Company is not obligated to issue any additional shares of its common stock, preferred stock or any additional options, warrants or other rights in or with respect to the unissued shares of such

stock or any other securities convertible into such stock. Section 4.2(a) of the Company Disclosure Letter is a list (the "Company Option List") setting forth the name of each holder of a Company Stock Option, the number of shares of Company Stock covered by each such option, the vesting schedule of each such option, the exercise price per share and the expiration date of each such option.

- (b) No bonds, debentures, notes or other indebtedness having the right to vote on any matters on which shareholders of Company may vote are issued and outstanding.
- (c) There are no shareholder agreements, voting trusts or other agreements or understandings to which the Company or any of its Subsidiaries is a party or of which the Company is otherwise aware with respect to the voting of the capital stock or other equity interest of the Company or any of its Subsidiaries.
- (d) No holder of securities in the Company or any of its Subsidiaries has any right to have such securities registered by the Company or any of its Subsidiaries, as the case may be.
- 4.3. <u>Subsidiaries</u>. Company has one wholly owned Subsidiary, namely, American Compensation Insurance Company ("<u>ACIC</u>"), a Minnesota corporation. ACIC has one wholly owned Subsidiary, namely, Bloomington Compensation Insurance Company ("<u>BCIC</u>"), a Minnesota corporation. None of Company, ACIC and BCIC has any Subsidiaries other than the Subsidiaries set out in this Section 4.3. The authorized capital stock of ACIC consists of 5,000,000 shares of common stock of which 1,000,000 shares are outstanding and held by the Company. The authorized capital stock of BCIC consists of 5,000,000 shares of common stock of which 666,667shares are outstanding and held by the Company. All such shares of Subsidiaries are validly issued, fully paid and non-assessable, and are clear of all Encumbrances. Other than Subsidiaries and securities held in its respective investment portfolio, none of the Company, ACIC or BCIC own any equity interest in any Person.
- 4.4. Financial Statements. Company has previously furnished to Parent a copy of the Financial Statements of Company. The Financial Statements of Company: (a) present fairly, in all material respects, the financial condition of Company as of the respective dates indicated and its statements of operations and changes in shareholders' equity and cash flows, for the respective periods then ended; and (b) have been prepared in accordance with GAAP consistently applied. The Company has delivered to Parent a copy of the statutory financial statements (including the annual reports filed with the Governmental Entities with jurisdiction over the Company and its Subsidiaries) for the years ended December 31, 2004, 2005 and 2006 and the financial statements for the three and six month period ended June 30, 2007. Each such statutory financial statement presents fairly and in accordance with Statutory Accounting Policies and practices prescribed or permitted by the appropriate regulatory agency of each state in which the statutory financial statements have or may be required to be filed, the financial position of each of ACIC and BCIC as of the date of each such referenced period. The amounts shown in the statutory financial statements reserves and liabilities for past and future insurance contract claims and expenses were computed (i) in all material respects in accordance with generally accepted actuarial standards consistently applied as in effect on such dates; (ii) in

compliance with applicable Laws, and (iii) consistent with actuarial assumptions used to compute corresponding statutory financial statements.

4.5. Reports and Filings.

- (a) Except as set forth in Section 4.5(a) of Company Disclosure Letter, Company has filed all required forms, reports, proxy statements, schedules, registration statements and other documents with the SEC since December 31, 2003 (the "Company SEC Documents"). As of their respective dates of filing with the SEC (or, if amended, supplemented or superseded by a filing prior to the date hereof, as of the date of such filing), the Company SEC Documents, including any financial statements or schedules included or incorporated by reference therein, complied in all material respects with the requirements of the Securities Act or the Exchange Act, as the case may be, and the rules and regulations of the SEC thereunder applicable to such Company SEC Documents, and none of the Company SEC Documents, including any financial statements or schedules included or incorporated by reference therein, when filed contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, except to the extent superseded or amended by a Company SEC Document filed subsequently and prior to the date hereof. The financial statements of Company included in the Company SEC Documents complied as to form, as of their respective dates of filing with the SEC, in all material respects with all applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto.
- (b) The Company has heretofore made, and hereafter will make, available to Parent a complete and correct copy of any amendments or modifications that are required to be filed with or submitted to the SEC but have not yet been filed with or submitted to the SEC to agreements, documents or other instruments that previously had been filed with or submitted to the SEC by the Company pursuant to the Exchange Act.
- (c) Each Company SEC Document containing financial statements that has been filed with or submitted to the SEC since July 31, 2002, was accompanied by the certifications required to be filed or submitted by the Company's chief executive officer and chief financial officer pursuant to the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), and at the time of filing or submission of each such certification, such certification was true and accurate.
- (d) Except as set forth in Section 4.5(d) of the Company Disclosure Letter, since December 31, 2003, neither the Company nor, to the Company's knowledge, any director, officer, employee, auditor, accountant or representative of the Company has received or otherwise had or obtained knowledge of any complaint, allegation, assertion or claim, whether written or oral, regarding the accounting or auditing practices, procedures, methodologies or methods of the Company or its respective internal accounting controls, including any complaint, allegation, assertion or claim that the Company has engaged in questionable accounting or auditing practices. No attorney representing the Company, whether or not employed by the Company, has reported evidence of a material violation of securities laws, breach of fiduciary

duty or similar violation by the Company or any of its officers, directors, employees or agents to the Company Board or any committee thereof or to any director or officer of the Company.

- 4.6. Authority of Company. The execution and delivery by Company of this Agreement and, subject to the requisite approval of the shareholders of Company of this Agreement and the Merger, the consummation of the transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the part of Company including, without limitation, the vote of the Board of Directors of Company (which vote was unanimous) approving this Agreement and the Merger. This Agreement is a valid and binding obligation of Company enforceable in accordance with its terms, except as the enforceability thereof may be limited by bankruptcy, liquidation, receivership, conservatorship, insolvency, fraudulent transfer, moratorium or other similar laws affecting the rights of creditors generally and by general equitable principles. Neither the execution and delivery by Company of this Agreement, the consummation of the transactions contemplated herein, nor compliance by Company with any of the provisions hereof, will: (a) conflict with or result in a breach of any provision of its Articles of Incorporation, as amended, or bylaws, as amended; (b) constitute a breach of or result in a default (or give rise to any rights of termination, cancellation or acceleration, or any right to acquire any securities or assets) under any of the terms, conditions or provisions of any Scheduled Contract; (c) result in the creation or imposition of any Encumbrance on any of the material properties or assets of Company; or (d) violate any material order, writ, injunction, decree, statute, rule or regulation applicable to Company or any of its properties or assets, except with respect to clauses (b), (c) and (d), for such violations, breaches, defaults or Encumbrances that would not, individually or in the aggregate, have a Material Adverse Effect. No consent of, approval of, notice to or filing with any Governmental Entity having jurisdiction over any aspect of the business or assets of Company is required in connection with the execution and delivery by Company of this Agreement or the consummation by Company of the Merger or the other transactions contemplated hereby or thereby, except (i) under the Exchange Act (including the filing of the Proxy Statement with the SEC); (ii) the necessary filings, applications and notices to and approvals and consents, if any, of the departments of the states charged with the regulation of insurance in the states in which the Company, ACIC or BCIC are licensed or authorized to do business; (iii) such other filings or notifications as may be required under federal or state securities law or the rules and regulations of NASDAQ Global Select market; (iv) such other consents, approvals, waivers, orders, authorizations, registrations, declarations and filings, which if not obtained or made would not, individually or in the aggregate, materially affect the ability of the Company to consummate the Merger, (v) applicable filings, notifications, approvals or consents under the HSR Act; and (vi) the filing of the Articles of Merger with the Secretary of State of the State of Minnesota and appropriate documents with relevant authorities of other states in which the Company is qualified to do business.
- 4.7. <u>Insurance</u>. Set forth in Section 4.7 of the Company Disclosure Letter is a list, as of the date hereof, of all policies of insurance carried and owned by Company and which are in force on the date hereof. No insurer under any such policy or bond has canceled or indicated an intention to cancel or not to renew any such policy or bond or generally disclaimed liability thereunder. Company is not in default under any such policy or bond that is material to the operations of Company and all material claims thereunder have been filed in a timely fashion.

- 4.8. <u>Personal Property</u>. Company has good title to all its properties and assets owned or stated to be owned by Company, free and clear of all Encumbrances except: (a) as set forth in the Financial Statements of Company; (b) for Encumbrances for current taxes not yet due; (c) for Encumbrances incurred in the ordinary course of business; or (d) for Encumbrances that are not substantial in character, amount or extent and that do not materially detract from the value, or interfere with present use, of the property subject thereto or affected thereby, or otherwise materially impair the conduct of business of Company.
- 4.9. <u>Real Estate</u>. None of Company, ACIC or BCIC own real property. Company, ACIC and BCIC each have a valid leasehold interest in all real property leased by the Company as described in Section 4.9 of the Company Disclosure Letter, free and clear of all Encumbrances, except (a) for rights of lessors, co-lessees or sublessees in such matters that are reflected in the lease; (b) for current taxes not yet due and payable; and (c) for such Encumbrances, if any, as do not materially detract from the value of or materially interfere with the present use of such property.
- 4.10. <u>Litigation</u>. Except as disclosed in the Company SEC Documents filed prior to the date of this Agreement or as set forth in the Company Disclosure Letter, there is no suit, action, investigation or proceeding (whether judicial, arbitral, administrative or other) pending or, to the knowledge of Company, threatened, against or affecting Company, ACIC or BCIC as to which there is a significant possibility of an adverse outcome that would, individually or in the aggregate, have a Material Adverse Effect, nor is there any judgment, decree, injunction, rule or order of any Governmental Entity outstanding against Company, ACIC or BCIC having or that would have, individually or in the aggregate, a Material Adverse Effect. To the Company's Knowledge, there are no judgments, decrees, stipulations or orders against Company or enjoining its directors, officers or employees in respect of, or the effect of which is to prohibit, any business practice or the acquisition of any property or the conduct of business in any area.
 - 4.11. Taxes. Subject to such exceptions as would not, individually or in the aggregate, have a Material Adverse Effect:
- (a) (i) All Tax Returns required to be filed by or on behalf of Company and its subsidiaries or the Affiliated Group(s) of which Company or a Subsidiary of Company is or was a member, have been duly and timely filed with the appropriate taxing authorities in all jurisdictions in which such Tax Returns are required to be filed (after giving effect to any valid extensions of time in which to make such filings), and all such Tax Returns were true, complete and correct; (ii) all Taxes due and payable by or on behalf of Company or its Subsidiaries, either directly, as part of an Affiliated Group Tax Return, or otherwise, have been fully and timely paid, except to the extent adequately reserved therefore in accordance with GAAP or applicable regulatory accounting principles or banking regulations consistently applied on the Company balance sheet, and adequate reserves or accruals for Taxes have been provided in the Company balance sheet with respect to any period through the date thereof for which Tax Returns have not yet been filed or for which Taxes are not yet due and owing; and (iii) no agreement, waiver or other document or arrangement extending or having the effect of extending the period for assessment or collection of Taxes (including, but not limited to, any applicable statute of limitation) has been executed or filed with any taxing authority by or on behalf of Company, its

Subsidiaries, or any Affiliated Group(s) of which the Company or any of its Subsidiaries is or was a member.

- (b) The Company and each of its Subsidiaries have complied with all applicable laws, rules and regulations relating to the payment and withholding of Taxes and has duly and timely withheld from any salaries, wages or other compensation or amounts paid to any employee, independent contractor, creditor, shareholder or other third party and has paid over to the appropriate taxing authorities all amounts required to be so withheld and paid over for all periods under all applicable Laws.
- (c) The Company and each of its Subsidiaries has furnished to Parent true and correct copies of (i) all income Tax Returns of Company relating to all taxable periods beginning after December 31, 2001; and (ii) any audit report issued within the last three years relating to any Taxes due from or with respect to Company and each of its Subsidiaries with respect to its income, assets or operations.
- (d) No written claim has been made by a taxing authority in a jurisdiction where Company or any of its Subsidiaries does not file an income, sales, use or franchise Tax Return such that Company or any of its Subsidiaries is or may be subject to taxation by that jurisdiction.
- (e) Except as set forth in Section 4.11(e) of the Company Disclosure Letter: (i) All deficiencies asserted or assessments made as a result of any examinations by any taxing authority of the Tax Returns of or covering or including Company, or any of its Subsidiaries, have been fully paid and, to the knowledge of the Company and each of its Subsidiaries, there are no other audits or investigations by any taxing authority in progress, nor has Company or any of its Subsidiaries received any written notice from any taxing authority that it intends to conduct such an audit or investigation; (ii) no requests for a ruling or a determination letter are pending with any taxing authority; and (iii) no issue has been raised in writing by any taxing authority in any current or prior examination that, by application of the same or similar principles, could reasonably be expected to result in a proposed deficiency against Company or any of its Subsidiaries for any subsequent taxable period. There are no liens for Taxes (other than Taxes not yet due or payable) upon any assets of the Company or any of its Subsidiaries.
- (f) Except as set forth in Section 4.11 (f) of the Company Disclosure Letter, neither Company nor any of its Subsidiaries are a party to any tax allocation, indemnification or sharing agreement (or similar agreement or arrangement), whether written or not written, pursuant to which it will have any obligation to make any payments after the Closing.
- (g) Neither Company nor any of its Subsidiaries has been a member of an Affiliated Group (other than a group whose common parent was Company).
- (h) Neither Company nor any of its Subsidiaries has liability for the Taxes of any person (other than the Company or any subsidiary, as applicable) under section 1.1502-6 of the Treasury Regulations (or any similar provision of state, local or foreign law), as a transferee or successor, by contract, or otherwise.

- (i) Neither Company nor any of its Subsidiaries has requests for rulings in respect of Taxes pending between Company and any taxing authority.
- (j) There is no contract, agreement, plan or arrangement covering any Person that, individually or collectively, could give rise to the payment of any amount that would not be deductible by Company or any of its Subsidiaries or their Affiliates by reason of Section 280G of the Code, or would constitute compensation in excess of the limitation set forth in Section 162(m) of the Code.
 - (k) There are no Encumbrances as a result of any due and unpaid Taxes upon any of the assets of Company or any of its Subsidiaries.
- (l) None of the Company or any of its Subsidiaries has engaged in a trade or business, had a permanent establishment (within the meaning of an applicable tax treaty or local law) or has otherwise become subject to Tax in a jurisdiction other than the country of its formation, and none of the Company or any of its Subsidiaries that are "U.S. persons" as that term is defined in Section 7701 of the Code has branches in any jurisdiction outside of the United States.
- (m) None of the Company or any of its Subsidiaries have participated, within the meaning of Treasury Regulation Section 1.6011-4(c), or have been a "material advisor" or "promoter" (as those terms are defined in "Section 6111 and 6112 of the Code and the Treasury Regulations promulgated thereunder) in (i) any "reportable transaction" within the meaning of Sections 6011, 6662A and 6707A of the Code and the Treasury Regulations promulgated thereunder, (ii) any "confidential corporate tax shelter" within the meaning of Section 6111 of the Code and the Treasury Regulations promulgated thereunder, or (iii) any "potentially abusive tax shelter" within the meaning of Section 6112 of the Code and the Treasury Regulations promulgated thereunder.
- (n) With respect to any reinsurance contracts to which the Company or any of its Subsidiaries is a party, no facts, circumstances or basis exists under which the IRS could make any reallocation, recharacterization or other adjustment under Section 845(a) of the Code, or make any adjustment arising from a determination that any reinsurance contract had or has a significant tax avoidance effect under Section 845(b) of the Code.
- (o) Within the past three years, neither the Company nor any of its Subsidiaries has been a "distributing corporation" or a "controlled corporation" in a distribution intended to qualify under Section 355(a) of the Code.

4.12. Compliance with Charter Provisions and Laws and Regulations(a) Company is not in default under or in breach or violation of (i) any provision of its Articles of Incorporation, as amended, or Bylaws, as amended, or (ii) any law, ordinance, rule or regulation promulgated by any Governmental Entity, except, with respect to this clause (ii), for such violations as would not have, individually or in the aggregate, a Material Adverse Effect. To the knowledge of Company, no investigation by any Governmental Entity with respect to Company is pending or threatened, other than, in each case, those the outcome of which, individually or in the aggregate, would not have a Material Adverse Effect.

(b) Each of the Company, ACIC and BCIC (i) in compliance with all Environmental Regulations; (ii) has not generated, used, stored, transported, disposed of, or arranged for the disposal of Hazardous Materials, with the exception of common cleaning materials which may have de minimus amounts of Hazardous Materials, if such materials were used, maintained, and disposed of in compliance with all Environmental Regulations, (iii) there are no Tanks on or about Company Property; (iv) there are no Hazardous Materials on, below or above the surface of, or migrating to or from Company Property; and (v) without limiting Section 4.10 hereof or the foregoing representations and warranties contained in clauses (i) through (v), as of the date of this Agreement, there is no claim, action, suit, or proceeding or notice thereof before any Governmental Entity pending against Company and there is no material outstanding judgment, order, writ, injunction, decree, or award against or affecting Company Property. For purposes of this Agreement, the term "Environmental Regulations" shall mean all Laws, licenses, permits, orders, approvals, plans, authorizations, concessions, franchises, and similar items, of all Governmental Entities and all applicable judicial, administrative, and regulatory decrees, judgments, and orders relating to the protection of human health or the environment, including, without limitation, those pertaining to reporting, licensing, permitting, investigation, and remediation of emissions, discharges, releases, or threatened releases of Hazardous Materials, chemical substances, pollutants, contaminants, or hazardous or toxic substances, materials or wastes whether solid, liquid, or gaseous in nature, into the air, surface water, groundwater, or land, or relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport, or handling of chemical substances, pollutants, contaminants, or hazardous or toxic substances, materials, or wastes, whether solid, liquid, or gaseous in nature and all requirements pertaining to the protection of the health and safety of employees or the public. "Company Property" shall mean real estate currently owned or leased by Company, ACIC or BCIC. "Tank" shall mean treatment or storage tanks, gas or oil wells and associated piping transportation devices. "Hazardous Materials" shall mean any substance: (1) the presence of which requires investigation or remediation under any federal, state or local statute, regulation, ordinance, order, action, policy or common law; (2) that is or becomes defined as a hazardous waste, hazardous substance, hazardous material, used oil, pollutant or contaminant under any federal, state or local statute, regulation, rule or ordinance or amendments thereto including, without limitation, the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. Section 9601, et seq.); the Resource Conservation and Recovery Act (42 U.S.C. Section 6901, et seq.); the Clean Air Act, as amended (42 U.S.C. Section 7401, et seq.); the Federal Water Pollution Control Act, as amended (33 U.S.C. Section 1251, et seq.); the Toxic Substances Control Act, as amended (15 U.S.C. Section 9601, et seq.); the Occupational Safety and Health Act, as amended (29 U.S.C. Section 651; the Emergency Planning and Community Right-to-Know Act of 1986 (42 U.S.C. Section 11001, et seq.); the Mine Safety and Health Act of 1977, as amended (30 U.S.C. Section 801, et seq.); the

Safe Drinking Water Act (42 U.S.C. Section 300f, *et seq.*); and all comparable state and local laws; (3) the presence of which causes or threatens to cause a nuisance, trespass or other common law tort upon real property or adjacent properties or poses or threatens to pose a hazard to the health or safety of persons or without limitation, that contains gasoline, diesel fuel or other petroleum hydrocarbons; or (4) polychlorinated biphenyls (PCBs), asbestos, lead-containing paints or urea formaldehyde foam insulation.

- 4.13. <u>Employees</u>. There are no controversies pending or, to the Company's knowledge, threatened between Company, ACIC or BCIC and any of their respective employees that could reasonably be expected to have a Material Adverse Effect. None of the Company, ACIC and BCIC is a party to any collective bargaining agreement with respect to any of their respective employees or any labor organization to which employees or any of them belong, and no union organizing effort is pending or threatened against the Company, ACIC or BCIC.
- 4.14. <u>Brokers and Finders</u>. Except for the obligation to KBW set forth in the KBW Agreement, a copy of which has been delivered to Parent, Company is not a party to or obligated under any agreement with any broker or finder relating to the transactions contemplated hereby, and neither the execution of this Agreement nor the consummation of the transactions provided for herein will result in any liability to any broker or finder.
- 4.15. <u>Scheduled Contracts</u>. Except as set forth in Section 4.15 of the Company Disclosure Letter or as disclosed in the Company SEC Documents (each item listed or required to be listed in such Company Disclosure Letter or the Company SEC Documents being referred to herein as a "Scheduled Contract" or "Insurance Contract" pursuant to Section 4.28 hereof), as of the date hereof, none of the Company, ACIC and BCIC is a party or otherwise subject to (other than purchase or sales orders entered into in the ordinary course):
- (a) any employment, deferred compensation, bonus or consulting contract that (i) has a remaining term, as of the date of this Agreement, of more than one year in length of obligation on the part of Company and is not terminable by Company within one year without penalty or (ii) requires payment by Company of \$100,000 or more per annum;
- (b) any advertising, brokerage, distributor, representative or agency relationship or contract requiring payment by Company of \$100,000 or more per annum;
- (c) any contract or agreement that restricts Company (or would restrict any Affiliate of Company or the Surviving Corporation (including Merger Sub and its Subsidiaries) after the Effective Time of the Merger) from competing in any line of business with any Person;
 - (d) any lease of real or personal property providing for annual lease payments by or to Company in excess of \$100,000 per annum;
 - (e) any license agreement granting any right to use or practice any right under Intellectual Property (whether as licensor or licensee);
- (f) any stock purchase, stock option, stock bonus, stock ownership, profit sharing, group insurance, bonus, deferred compensation, severance pay, pension, retirement,

savings or other incentive, welfare or employment plan or material agreement providing benefits to any present or former employees, officers or directors of Company;

- (g) any agreement to acquire equipment or any commitment to make capital expenditures of \$100,000 or more;
- (h) any agreement for the sale of any material property or assets in which Company has an ownership interest or for the grant of any Encumbrance on any such property or asset, except for investment portfolio transactions in the ordinary course of business;
 - (i) any agreement for the borrowing of any money and any guaranty agreement;
 - (j) any partnership or joint venture agreement;
- (k) any material agreement that would be terminable other than by Company as a result of the consummation of the transactions contemplated by this Agreement; or
- (l) other than agreements entered into in the ordinary course of business, any other agreement of any other kind that involves future payments or receipts or performances of services or delivery of items requiring payment of \$100,000 or more to or by Company.

Complete copies of all Scheduled Contracts, including all amendments and supplements thereto, have been delivered or made available to Parent.

- 4.16. <u>Performance of Obligations</u>. Company has performed in all respects all of the obligations required to be performed by it to date and is not in default under or in breach of any term or provision of any Scheduled Contract to which it is a party, is subject or is otherwise bound, and no event has occurred that, with the giving of notice or the passage of time or both, would constitute such default or breach, except where such failure of performance, breach or default would not individually or in the aggregate have a Material Adverse Effect. Except as set forth in Section 4.16 of the Company Disclosure Letter, to Company's knowledge, no other party to any Scheduled Contract is in default thereunder.
- 4.17. <u>Certain Material Changes</u>. Except as specifically required, permitted or effected by this Agreement, or as disclosed in the Company SEC Documents, since December 31, 2006, there has not been, occurred or arisen any of the following (whether or not in the ordinary course of business unless otherwise indicated):
 - (a) any change in methods of accounting or accounting practices, business, or manner of conducting business, of Company;
 - (b) any other event or development that has had, individually or in the aggregate, a Material Adverse Effect;
 - (c) any material damage, destruction or other casualty loss (whether or not covered by insurance);

- (d) any amendment, modification or termination of any existing, or entry into any new, material contract or permit;
- (e) any disposition by Company of a material asset;
- (f) or any direct or indirect redemption, purchase or other acquisition by Company of any equity securities or any declaration, setting aside or payment of any dividend or other distribution on or in respect of Company Stock whether consisting of money, other personal property, real property or other things of value (except for dividends permitted by Section 6.1(b)).
- 4.18. <u>Licenses and Permits</u>. Each of Company, ACIC and BCIC has all licenses and permits that are necessary for the conduct of its respective business, and such licenses are in full force and effect in all material respects. The respective properties, assets, operations and businesses of Company, ACIC and BCIC are and have been maintained and conducted, in all material respects, in compliance with all such applicable licenses and permits. No proceeding is pending or to the knowledge of the Company, threatened by any Governmental Entity that seeks to revoke or limit any such licenses or permits.
- 4.19. <u>Undisclosed Liabilities</u>. Except for liabilities or obligations that do not individually or in the aggregate have a Material Adverse Effect, none of the Company, ACIC and BCIC has any liabilities or obligations of any nature, either accrued, contingent or otherwise, whether known or unknown, and whether due or to become due, except those (a) reflected or disclosed in the Financial Statements of Company or the statutory financial statements of ACIC or BCIC; (b) incurred subsequent to June 30, 2007 in the ordinary course of business consistent with past practices; or (c) disclosed in the Company Disclosure Letter or Company SEC Documents.

4.20. Employee Benefit Plans.

(a) Company has previously made available to Parent copies of current documents constituting of each "employee benefit plan," as defined in Section 3(3) of ERISA, of which Company or any member of the same controlled group of corporations, trades or businesses as Company within the meaning of Section 4001(a)(14) of ERISA ("ERISA Affiliates") is a sponsor or participating employer or as to which Company or any of its ERISA Affiliates makes contributions or is required to make contributions and which is subject to any provision of ERISA and covers any current or former employee, director, agent or independent contractor, of Company or any of its ERISA Affiliates, together with all amendments thereto, all currently effective and related summary plan descriptions with respect to plans subject to Section 401(a) of the Code, the determination letter from the IRS, the annual reports for the most recent three years (Form 5500 including, if applicable, Schedule B thereto, and Form 11-K, if applicable) and a summary of modifications prepared in connection with any such plan. Such plans are hereinafter referred to collectively as the "Employee Plans," and are listed in Section 4.20(a) of the Company Disclosure Letter. No Employee Plan is a "multiemployer plan" within the meaning of Section 3(37) of ERISA, and neither the Company, its Subsidiaries nor any ERISA Affiliate has at any time sponsored or contributed to, or had any liability or obligation in respect of, any "multiemployer plan" for the five years preceding the date of this Agreement.

Neither the Company nor any of its Subsidiaries has any obligation to provide health, life or other benefits of any kind to any retired or former employee of either the Company of any of its Subsidiaries. Each Employee Plan that is intended to be qualified in form and operation under Section 401(a) of the Code has received a favorable determination opinion or notification letter from the IRS and the associated trust for each such Employee Plan is exempt from tax under Section 501(a) of the Code and Company knows of no fact that would adversely affect the qualified status of any such Employee Plan. No event has occurred that will subject such Employee Plans to any material tax under Section 511 of the Code. All amendments required to bring each Employee Plan into conformity with all of the applicable provisions of ERISA, the Code and all other applicable laws have been made, except to the extent that such amendments that would retroactively cover any period prior to the Effective Time of the Merger are not required to be adopted prior to the Effective Time of the Merger.

- (b) Company has previously made available to Parent copies or descriptions of each written or unwritten plan or arrangement maintained or otherwise contributed to, or as to which the Company or any ERISA Affiliate is obligated, by Company or any of its ERISA Affiliates that is not an Employee Plan and that (exclusive of base salary and base wages and any benefit required solely under the law of any state) provides for any form of current or deferred compensation, bonus, stock option, stock awards, stock-based compensation or other forms of incentive compensation, or insurance, profit sharing, benefit, retirement, group health, disability, workers' compensation, welfare or similar plan or arrangement for the benefit of any particular or class of current or former employees, directors, agents or independent contractors of Company or any of its ERISA Affiliates. Such plans and arrangements are hereinafter collectively referred to as "Benefit Arrangements"), and are listed in Section 4.20(b) of the Company Disclosure Letter. Except as set forth in the Company Disclosure Letter, there has been no increase in the compensation of or benefits payable to any senior executive employee of Company since June 30, 2007, nor any employment, severance or similar contract entered into with any such employee, nor any amendment to any such contract, since June 30, 2007.
- (c) With respect to all Employee Plans and Benefit Arrangements, Company and its ERISA Affiliates are in compliance with the terms of such Employee Plans and Benefit Arrangements, the requirements prescribed by any and all statutes, governmental or court orders, or governmental rules or regulations currently in effect, including but not limited to ERISA and the Code, applicable to such plans or arrangements. All government reports and filings required by law have been properly and timely filed and all information required to be distributed to participants or beneficiaries has been distributed with respect to each Employee Plan. There is no pending or, to the Company's knowledge, threatened legal action, proceeding or investigation against or involving any Employee Plan or Benefit Arrangement, other than routine claims for benefits. No "prohibited transaction," as defined in Section 406 of ERISA or Section 4975 of the Code, has occurred with respect to any Employee Plan that could subject the Company or any person for whom the Company has an obligation to indemnity to liability under Title I of ERISA or to the imposition of tax under Section 4975 of the Code. No Employee Plan is subject to Title IV or Part 3 of Subtitle B of Title I of ERISA or Section 412 of the Code. No "reportable event" as defined in ERISA has occurred with respect to any of the Employee Plans. All contributions required to be made to each of the Employee Plans or Benefit Arrangement under the terms of the Employee Plan, Benefit Arrangement and ERISA, the Code or any other applicable Law have been timely made.

- 4.21. <u>Corporate Records</u>. The minute books of Company and each Subsidiary accurately reflect all material corporate actions taken since December 31, 2003 to this date by the respective shareholders, Board of Directors and committees of Company or Subsidiary, as the case may be.
- 4.22. Accounting Records and Internal Controls. The Company has established and maintains a system of internal control over financial reporting regarding the reliability of financial reporting and the preparation of its consolidated financial statements in accordance with generally accepted accounting principles in the United States, including policies and procedures that (i) pertain to the maintenance or records that, in reasonable detail, accurately and fairly reflect the transactions and disposition of assets of the Company and its Subsidiaries, (ii) provide reasonable assurance that the Company and its transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, and that receipts and expenditures of the Company and its Subsidiaries are being made only in accordance with appropriate authorization of management and the Board of Directors of the Company, and (iii) provide reasonable assurance regarding prevention or detection of unauthorized acquisition, use or disposition of the assets of the Company and its Subsidiaries. The Company has disclosed, based on its most recent evaluation of internal control over financial reporting to the Company's independent auditors and the audit committee of the Board of Directors of the Company all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting that are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information and any fraud, whether or not material, that involves Company management or other employees who have a significant role in the Company's internal control over financial reporting. The Company has established and maintains disclosure controls and procedures to ensure that all material information relating to the Company and its Subsidiaries required to be disclosed by the Company in the reports that its files with the SEC under the Exchange Act and that such material information is accumulated and communicated to the Company's management for inclusion in such SEC filings.
- 4.23. <u>Vote Required</u>. The affirmative vote of the holders of a majority of the outstanding shares of Company Stock to adopt this Agreement is the only vote of the holders of any class or series of Company capital stock necessary to approve and adopt this Agreement and the transactions contemplated hereby (including the Merger).
- 4.24. <u>Disclosure Documents and Applications</u>. None of the information supplied or to be supplied by Company in writing ("<u>Company Supplied Information</u>") for inclusion in any documents to be filed with the SEC or any other Governmental Entity in connection with the transactions contemplated in this Agreement, will, at the respective times such documents are filed or become effective, or with respect to the Proxy Statement when mailed, with respect to the Company Supplied Information, contain any untrue statement of a material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

4.25. <u>Intellectual Property</u>

- (a) As used herein, the term "<u>Intellectual Property</u>" means all intellectual property rights arising under the laws of the United States or any other jurisdiction with respect to the following: (i) trade names, trademarks and service marks (registered and unregistered), domain names and applications to register any of the foregoing (collectively, "<u>Marks</u>"); (ii) patents and patent applications (collectively, "<u>Patents</u>"); (iii) copyrights and registrations and applications therefore (collectively, "<u>Copyrights</u>"); and (iv) know-how, inventions, discoveries, methods, processes, technical data, specifications, customer lists, in each case that derives economic value (actual or potential) from not being generally known to other persons who can obtain economic value from its disclosure, but excluding any Copyrights or Patents that cover or protect any of the foregoing (collectively, "<u>Trade Secrets</u>").
- (b) The Company Disclosure Letter sets forth an accurate and complete list of all registered Marks and applications for registration of Marks owned by the Company (collectively, "Company Registered Marks"), the Company Disclosure Letter also sets forth an accurate and complete list of all Patents owned by the Company (collectively, the "Company Patents" and, together with the Company Registered Marks, the "Company Registered IP"). Except as set forth in the Company Disclosure Letter, no Company Registered IP has been or is now involved in any interference, reissue, reexamination, opposition or cancellation proceeding and, to the knowledge of the Company, no such action is or has been threatened with respect to any of the Company Registered IP. Except as set forth in the Company Disclosure Letter, the Company Registered IP relating to the Company's existing products and products currently under development is valid, subsisting and, to the knowledge of the Company, enforceable, and no notice or claim challenging the validity or enforceability or alleging the misuse of any of the Company Registered IP has been received by the Company. Except as set forth in the Company Disclosure Letter, and other than abandoned patent applications or closed patent application files, (i) the Company has not taken any action or failed to take any action that could reasonably be expected to result in the abandonment, cancellation, forfeiture, relinquishment, invalidation or unenforceability of any of the Company Registered IP, and (ii) all filing, examination, issuance, post registration and maintenance fees, annuities and the like associated with or required with respect to any of the Company Registered IP have been timely paid.
- (c) The Company owns or possesses adequate licenses or other valid rights to use, all of the Intellectual Property that is necessary for the conduct of the Company's businesses as currently conducted. Except as set forth in the Company Disclosure Letter, none of the Intellectual Property owned by the Company is subject to any outstanding order, judgment, or stipulation restricting the use thereof by the Company.
- (d) The rights licensed under each agreement granting to the Company any material right or license under or with respect to any Intellectual Property owned by a third party shall be exercisable by the Surviving Corporation on and after the Closing to the same extent as by the Company prior to the Closing. No loss or expiration of any such agreement is pending or reasonably foreseeable or, to the knowledge of the Company, threatened. Any and all license fees, royalties, or other amounts due with respect to any such licensed Intellectual Property licensed have been paid in full. The Company has not granted to any third party any exclusive

rights under any Intellectual Property owned by the Company or otherwise granted any rights under such Intellectual Property outside the ordinary course of business.

- (e) The Company has taken reasonable steps to protect its rights in the Intellectual Property owned by the Company and maintain the confidentiality of all material Trade Secrets of the Company.
- (f) The Intellectual Property owned by or validly licensed to the Company or its Subsidiaries constitutes all the material Intellectual Property rights necessary for the conduct of their respective businesses as currently conducted and contemplated to be conducted.
- (g) To the knowledge of the Company, none of the products or services distributed, sold or offered by the Company, nor any technology, materials or Intellectual Property used, sold, distributed or otherwise commercially exploited by or for the Company, infringes upon, misappropriates or violates any Intellectual Property of any third party or constitutes unfair competition or trade practices under the laws of any jurisdiction and, to the knowledge of the Company, the Company has not received any notice or claim asserting or suggesting in writing that any such infringement, misappropriation, violation, or dilution, unfair competition or trade practices has occurred. To the knowledge of the Company, except as set forth in the Company Disclosure Letter, (i) no third party is misappropriating or infringing any material Intellectual Property owned by the Company; and (ii) no third party has made any unauthorized disclosure of any material Trade Secrets of the Company.
- 4.26. <u>State Takeover Laws</u>. The Board of Directors of Company has approved this Agreement and the transactions contemplated hereby, and taken such other actions, and such actions are sufficient, to render inapplicable to this Agreement and the transactions contemplated hereby, including, without limitation, the Merger, the Minnesota Control Share Acquisition Act, the Minnesota Business Combination Act, all applicable state takeover statutes and any similar "takeover" or "interested shareholder" Laws.
- 4.27. Opinion of KBW. Company has received the opinion of KBW dated as of the date hereof, to the effect that, based upon and subject to the matters set forth in the opinion, the Merger Consideration is fair from a financial point of view to the holders of the Company Stock.

4.28. Insurance Matters.

(a) <u>Insurance Subsidiaries and Insurance Contracts</u>. Each of ACIC and BCIC: (i) is duly licensed or authorized as an insurance company in Minnesota; (ii) duly licensed or authorized or otherwise eligible to act as an insurance company in each other jurisdiction where it is required to be so licensed, authorized or eligible; and (iii) duly licensed, authorized or eligible in Minnesota and each other applicable jurisdiction to write each line of business reported as being written in each Subsidiaries' statutory financial statements. Each jurisdiction in which each Subsidiary is licensed or authorized under (i) through (iii) of the foregoing sentence is set forth in Section 4.28 of the Company Disclosure Letter, to the extent required by applicable Laws, each insurance contract issued or distributed by each Subsidiary in any jurisdiction since January 1, 2002, is to the extent required by applicable Laws, on a form

approved by the applicable insurance department or has been filed and not objected to by such insurance department within the period provided for such objections, and such form complies with applicable Laws.

(b) <u>Underwriting Management and Administration Agreements</u>. All underwriting management and administration agreements entered into by the Company and its Subsidiaries as now in force are, to the extent required by applicable Laws, in forms acceptable to the applicable insurance departments of applicable jurisdictions (or have been submitted for approval which is pending, or have been filed and not objected to by such insurance departments within the period provided for objection).

(c) Reinsurance and Retrocession.

- (i) Each insurance contract, treaty or arrangement (including any facilitative agreements, indemnity agreements, or terminated or expired treaty or agreement under which there remains any outstanding material liability with respect to paid or unpaid case reserves regarding ceding or assumption of reinsurance, coinsurance, excess insurance, or retrocessions) ("Reinsurance Contracts") to which the Company's Subsidiaries are a party or by or to which any of them are bound or subject, as each such Reinsurance Contract may have been amended, modified or supplemented is a valid and binding obligation of the parties thereto, is in full force and effect and is enforceable in accordance with its terms, and each such Reinsurance Contract is listed on Section 4.28(c) of the Company Disclosure Letter. Neither the Company's Subsidiaries nor, to the Company's Knowledge, any other party thereto, is in default in any material respect, nor to Company's Knowledge is any default threatened by such party, with respect to any such Reinsurance Contract.
- (ii) Each of the Company's Subsidiaries is entitled under applicable law to take full credit in its statutory financial statements for all amounts recoverable by it pursuant to any Reinsurance Contract, and all such amounts recoverable have been properly recorded in the books and records of account of the Company and its Subsidiaries and are properly reflected in the statutory financial statements. To the Company's Knowledge, all such amounts recoverable by the Company or any of its Subsidiaries are fully collectible in due course. Neither the Company nor any of its Subsidiaries has received notice that any other party to any Reinsurance Contracts intends not to perform under any such Reinsurance Contracts, and, to the Company's Knowledge, the financial condition of each other party to each Reinsurance Contract pursuant to which its Subsidiaries have ceded any premiums is not impaired to the extent that a default thereunder is reasonably anticipated.
- (d) <u>Insurance Ratings</u>. As of the date of this Agreement the Company's Subsidiaries have been assigned a "B++" insurer financial strength rating with a positive outlook by A. M. Best Co. Except as set forth on Section 4.28 of the Company Disclosure Letter, since December 31, 2005 the Company has not been downgraded and, to the Knowledge of the Company as of the date of this Agreement, A. M. Best Co. has not indicated that it intends to lower its rating or put the Company on an "under review" status.

- (e) Agents and Brokers. Except as set forth on Section 4.28 (e) of the Company Disclosure Letter no single agent broker, intermediary, manager or producer employed or engaged by the Company or any of its Subsidiaries ("Producer") generated more than ten percent (10%) of the aggregate gross written premium of the Company or any of its Subsidiaries during either of the years ended December 31, 2005 or December 31, 2006. To the Company's Knowledge, except as set forth in Schedule 4.28(e) of the Company Disclosure Letter, each Producer complies in all material respects with applicable Laws regarding such Producer's authority to engage in the type of insurance activities in which such Producer is engaged and each Producer is duly licensed (including without limitation, the marketing, sale or issuance of any insurance contracts) in each jurisdiction in which such Producer places or sells insurance contracts on behalf of the Company or its Subsidiaries, and each such Producer is duly authorized and appointed by the applicable Subsidiary pursuant to applicable Laws in all material respects. All contracts between any Producer who accounted for more than ten (10%) of the aggregate gross written premiums of the Company or its Subsidiaries during the year ended December 31, 2006 have given or been given written notice of termination or, to the Company's Knowledge, threatened or been threatened with termination or threatened or been threatened with a substantial reduction in the amount of premiums to be written by such Person on behalf of the Company or its Subsidiaries. There is no dispute pending or, to the Company's Knowledge, threatened against the Company or any of its Subsidiaries by any Producer.
- (f) <u>Agreements with Regulators</u>. Except as set forth in Section 4.28 (f) of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries is a party to any written agreement, consent decree or memorandum of understanding with, or a party, to any commitment letter or similar undertaking to, or is subject to any cease-and-desist or other order or directive by, or has adopted any policies, procedures, or board resolutions at the request of, any Governmental Entity which restricts the conduct of the business of Company or any of its Subsidiaries, or relates to the Company's or any of its Subsidiaries' capital adequacy or risk management policies, nor has the Company or any of its Subsidiaries been advised in writing by any Governmental Entity that is it contemplating any such undertakings.
- 4.29. <u>Restrictions on Business Activities</u>. Except as set forth in Section 4.29 of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries is party to or bound by any contract containing any covenant limiting in any material respect the right of the Company or any of its Subsidiaries to engage or compete in any line of business.
- 4.30. No Additional Representations. The Company does not make, and has not made, any representations or warranties relating to the Company or the business of the Company or otherwise in connection with the transactions contemplated hereby other than those expressly set forth in this Agreement that are made by the Company. Without limiting the generality of the foregoing, the Company has not made, and shall not be deemed to have made, any representations and warranties in any presentation of the business of the Company in connection with the transactions contemplated hereby and, accordingly, no statement made in any such presentation shall be deemed a representation or warranty hereunder or otherwise. It is understood that any cost estimates, projections or other predictions, any data, any financial information or any memoranda or offering materials or presentations are not and shall not be deemed to be or to include representations or warranties of the Company. No person has been authorized by the Company to make any representation or warranty relating to the Company, the

business of the Company or otherwise in connection with the transactions contemplated hereby and, if made, such representation or warranty must not be relied upon as having been authorized by the Company.

ARTICLE 5.

REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Company as follows:

- 5.1. <u>Incorporation, Standing and Power</u>. Parent has been duly organized, is validly existing and in good standing as a corporation under the laws of Delaware. Merger Sub has been duly organized, is validly existing and in good standing as a corporation under the laws of the State of Minnesota. Merger Sub has conducted no business or operations and has no material liabilities other than its obligations under this Agreement.
- 5.2. Authority. The execution and delivery by Parent of this Agreement, and the consummation of the transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the part of Parent. The execution and delivery by Merger Sub of this Agreement, and the consummation of the transactions contemplated hereby, have been duly and validly authorized by all necessary corporate action on the part of Merger Sub. This Agreement is a valid and binding obligation of Parent and Merger Sub, in each case enforceable in accordance with its terms, except as the enforceability thereof may be limited by bankruptcy, liquidation, receivership, conservatorship, insolvency, fraudulent conveyance, moratorium or other similar laws affecting the rights of creditors generally and by general equitable principles. Neither the execution and delivery by Parent or Merger Sub of this Agreement, the consummation of the transactions contemplated herein, nor compliance by Parent and Merger Sub with any of the provisions hereof, will: (a) conflict with or result in a breach of any provision of its respective governing documents; (b) constitute a breach of or result in a default (or give rise to any rights of termination, cancellation or acceleration, or any right to acquire any securities or assets) under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, franchise, license, permit, agreement or other instrument or obligation to which Parent or any Subsidiary of Parent is a party, or by which Parent or any Subsidiary of Parent or any of its properties or assets is bound (except as would not be reasonably likely to have a material adverse effect on the ability of Parent and Merger Sub to consummate the transactions contemplated by this Agreement); or (c) violate any order, writ, injunction, decree, statute, rule or regulation applicable to Parent or any Subsidiary of Parent or any of their respective properties or assets. No consent of, approval of, notice to or filing with any Governmental Entity having jurisdiction over any aspect of the business or assets of Parent or any of its Subsidiaries, and no consent of, approval of or notice to any other Person, is required in connection with the execution and delivery by Parent or Merger Sub of this Agreement, or the consummation by Parent and Merger Sub of the Merger or the transactions contemplated hereby, except (i) applicable filings, notifications, approvals or consents under the HSR Act and applicable insurance departments; and (ii) the filing of the Articles of Merger with the Secretary of State of the State of Minnesota.

- 5.3. <u>Financing</u>. Parent has available sufficient cash or other liquid assets or financial resources which may be used to fund the Merger and perform its other obligations hereunder. Parent's ability to consummate the transactions contemplated by this Agreement is not contingent on raising any equity capital, obtaining financing therefore, consent of any lender or any other matter.
- 5.4. <u>Litigation</u>. No claim, action, proceeding or investigation is pending or, to the knowledge of Parent, threatened, that seeks to delay or prevent the consummation of, or that would be reasonably likely to materially adversely affect Parent's or Merger Sub's ability to consummate the transactions contemplated by this Agreement.
- 5.5. Ownership of Merger Sub. Merger Sub is a direct wholly owned subsidiary of Parent. Merger Sub has not conducted any activities other than in connection with the organization of Merger Sub, the negotiation and execution of this Agreement and the consummation of the transactions contemplated hereby. Merger Sub has no Subsidiaries.
- 5.6. <u>Facts Affecting Regulatory Approvals</u>. To the knowledge of Parent, there is no fact, event or condition applicable to Parent or any of its Subsidiaries which will, or reasonably could be expected to, adversely affect the likelihood of promptly securing the requisite approvals or consents of any Governmental Entity to the Merger.
- 5.7. Accuracy of Information Furnished for Company Proxy Statement. None of the information supplied or to be supplied by Parent in writing ("Parent Supplied Information") for inclusion in any documents to be filed by Company with the SEC or any other Governmental Entity in connection with the transactions contemplated in this Agreement, will, at the respective times such documents are filed or become effective, or with respect to the Proxy Statement when mailed, with respect to the Parent Supplied Information, contain any untrue statement of a material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

ARTICLE 6.

COVENANTS OF COMPANY PENDING EFFECTIVE TIME OF THE MERGER

Company covenants and agrees with Parent and Merger Sub as follows:

- 6.1. <u>Limitation on Conduct Prior to Effective Time of the Merger</u>. Between the date hereof and the earlier of the Effective Time of the Merger or the termination of the Agreement, except as contemplated by this Agreement and subject to applicable Laws, Company agrees to conduct its business in the ordinary course in substantially the manner heretofore conducted, and Company shall not, without the prior written consent of Parent; which consent shall not be unreasonably withheld or delayed:
- (a) issue, sell or grant any Company Stock (except pursuant to the exercise of Company Stock Options outstanding as of the date hereof), any other securities (including long term debt) of Company, or any rights, stock appreciation rights, options or securities to acquire

any Company Stock, or any other securities (including long term debt) of Company or enter into any agreements to take any such actions, except that the Company may continue to permit employees that are participants in the 1995 Employee Stock Purchase Plan and Trust (the "ESPP") as of the date hereof, to continue to participate in the ESPP until the earlier of the Effective Time of the Merger or December 31, 2007, subject to the rules and restrictions of the ESPP, and the Company may issue Company Stock upon the exercise of Company Stock Options pursuant to the ESPP, provided that no employee currently participating in the ESPP may increase his or her payroll deductions, and the Company may not issue more shares than the shares specified in Section 4.2(a) of the Company Disclosure Letter;

- (b) (i) declare, set aside or pay any dividend or make any other distribution upon any of the capital stock of Company, or (ii) split, combine or reclassify any shares of capital stock or other securities of Company;
- (c) purchase, redeem or otherwise acquire any capital stock or other securities of Company or any rights, options, or securities to acquire any capital stock or other securities of Company (other than the issuance of Company Stock upon the exercise of Company Stock Options that are outstanding as of the date hereof in accordance with their present terms);
 - (d) except as may be required to effect the transactions contemplated herein, amend its Articles of Incorporation or bylaws;
- (e) grant any general or uniform increase in the rate of pay of employees or employee benefits other than in the ordinary course of business consistent with past practices;
- (f) except as provided in the Company Disclosure Letter and in Section 12.1, grant any increase in salary, incentive compensation or employee benefits or pay any bonus to any Person or voluntarily accelerate the vesting of any employee benefits, other than payments of bonuses consistent with past practice pursuant to plans in effect on the date hereof and disclosed in the Company Disclosure Letter, increases in salary consistent with past practice to Persons eligible for such salary increases on the regularly scheduled review dates of their employment, provided that the percentage increase in salaries for all such Persons shall not exceed four percent on average and payment of retention bonuses to employees deemed necessary by management;
- (g) make any capital expenditure or commitments with respect thereto in excess of \$100,000 with respect to any item or project or in the aggregate with respect to any related items or projects, except for capital expenditures described in the Company Disclosure letter and ordinary repairs, renewals and replacements;
- (h) compromise or otherwise settle or adjust any assertion or claim of a deficiency in taxes (or interest thereon or penalties in connection therewith), extend the statute of limitations with any tax authority or file any pleading in court in any tax litigation or any appeal from an asserted deficiency, or file or amend any federal, foreign, state or local tax return, or make any tax election that is inconsistent with Company's current tax election practices;

- (i) change or make any tax elections or its tax or accounting policies and procedures or any method or period of accounting unless required by GAAP, Statutory Accounting Policies or a Governmental Entity;
- (j) grant or commit to grant any extension of credit or amend the terms of any such credit outstanding on the date hereof to any executive officer, director or holder of 10% or more of the outstanding Company Stock, or any Affiliate of such Person;
 - (k) close or relocate any principal offices at which business is conducted or open any new principal offices;
- (l) except as provided in the Company Disclosure Letter, in Section 12.1 or elsewhere in this Section 6.1, adopt or enter into any new employment agreement with any executive level employee or other employee benefit plan or arrangement or amend or modify any employment agreement or employee benefit plan or arrangement of any such type except for such amendments as are required by applicable Laws;
- (m) initiate, solicit or knowingly encourage (including by way of furnishing information or assistance), or take any other action to facilitate, any inquiries or the making of any proposal which constitutes, or would reasonably be expected to lead to, any Competing Transaction (as such term is defined below), or negotiate or have any discussions with any person in furtherance of such inquiries or to obtain a Competing Transaction, or agree to or endorse any Competing Transaction, or approve or recommend, or execute or enter into, any letter of intent, agreement in principle, merger agreement, asset purchase or share exchange or issuance agreement, option agreement, or other similar agreement related to any Competing Transaction or agree to do any of the foregoing, or authorize any of its officers, directors or employees or any investment banker, financial advisor, attorney, accountant or any other representative retained by it or any of its Affiliates (the "Representatives") to take any such action, and will cause the Representatives not to take any such action, and Company shall notify Parent of all of the relevant details relating to all inquiries and proposals which it may receive relating to any of such matters. For purposes of this Agreement, "Competing Transaction" shall mean any of the following involving Company and any Person other than Parent or any of its Affiliates: any merger, consolidation, share exchange or other business combination; a sale, lease, exchange, mortgage, pledge, transfer or other disposition of assets of Company representing 15% or more of the consolidated assets of Company; a sale of shares of capital stock (or securities convertible or exchangeable into or otherwise evidencing, or any agreement or instrument evidencing, the right to acquire capital stock), representing 15% or more of the voting power of Company; or a tender offer or exchange offer for at least 15% of the outstanding shares of Company. Company will immediately cease and cause to be terminated any existing activities, discussions or negotiations with any parties (other than Parent) conducted heretofore with respect to any of the foregoing. Company shall take the necessary steps to inform promptly the appropriate individuals or entities referred to above of the obligations undertaken in this Section. Company shall notify Parent within 48 hours of the receipt of any such inquiries, proposals or offers, the request for any such information, or the initiation or continuation of any such negotiations or discussions which are sought to be initiated or continued with Company. Notwithstanding any other provision in this Section 6.1(m), prior to the duly convened meeting of the shareholders of the Company required by Section 6.6, and subject to compliance with

other terms of this Section 6.1(m), and to first entering into a confidentiality agreement having confidentiality provisions that are no less favorable to Company than those contained in the Confidentiality Agreement, the Board of Directors of Company shall be permitted to engage in discussions or negotiations with, or provide any nonpublic information or data to, any Person in response to an unsolicited bona fide written proposal for a Competing Transaction by such Person first made after the date hereof which the Board of Directors of Company concludes in good faith (after consultation with its financial advisor) constitutes or is reasonably likely to result in a Superior Proposal (as defined below), or to recommend such Superior Proposal to the holders of Company Stock, if and only to the extent that the Board of Directors of Company reasonably determines in good faith (after consultation with outside legal counsel) that failure to do so would be inconsistent with its fiduciary duties under applicable law; provided, that Company and the Board of Directors of Company shall have given Parent at least 48 hours prior notice of its intent to do so before taking any such action; provided, further, that Company and the Board of Directors of Company shall keep Parent informed of the status and terms of any such proposals, offers, discussions or negotiations on a current basis. Parent has the right to meet any such proposal for a Competing Transaction at any time prior to the Board of Directors of Company making a determination that the Competing Transaction is a Superior Proposal. For purposes of this Agreement, "Superior Proposal" shall mean a bona fide written proposal for a Competing Transaction which the Board of Directors concludes in good faith, after consultation with its financial advisor and its legal advisors, taking into account all legal, financial, regulatory and other aspects of the proposal and the Person making the proposal is more favorable to the Company's shareholders from a financial point of view, than the transactions contemplated by this Agreement. Provided, that, for purposes of this definition of "Superior Proposal" the term Competing Transaction shall have the meaning assigned to such term in this Section 6.1(m), except that the reference to "15% or more" in the definition of Competing Transaction shall be deemed to be a reference to "a majority." Nothing in this Section 6.1(m) shall prohibit Company or its Board of Directors from taking and disclosing to the Company shareholders a position with respect to a Competing Transaction to the extent required under the Exchange Act, or from making such disclosure to the Company shareholders which, after consultation with outside counsel, the Board determines is otherwise required under applicable law;

- (n) shall use its commercially reasonable best efforts to not incur expenses related to the execution, delivery and performance of this Agreement in excess of \$2 million;
 - (o) grant any Person a power of attorney or similar authority;
- (p) make any investment by purchase of stock or securities, contributions to capital, property transfers or otherwise in any other Person, except for federal funds, obligations of the United States Treasury or an agency of the United States Government the obligations of which are entitled to or implied to have the full faith and credit of the United States government and which have an original maturity not in excess of one year, bank qualified investment grade municipal bonds, in any case, in the ordinary course of business consistent with past practices;
- (q) amend or modify any Scheduled Contract or enter into any agreement or contract that would be required to be a Scheduled Contract under Section 4.15; <u>provided</u>, that Company may renew an existing Scheduled Contract in the ordinary course of business on substantially equivalent terms;

- (r) sell, transfer, mortgage, encumber or otherwise dispose of any material assets or release or waive any material claim;
- (s) take any action which would or could reasonably be expected to (i) adversely affect the ability of Parent or Company to obtain any necessary approval of any Governmental Entity required for the transactions contemplated hereby; (ii) adversely affect Company's ability to perform its covenants and agreements under this Agreement; or (iii) result in any of the conditions to the performance of Parent's or Company's obligations hereunder, as set forth in Articles 9, 10 or 11 herein not being satisfied;
 - (t) make any special or extraordinary distributions or payments to any Person;
- (u) settle any material claim, action or proceeding involving any material liability for monetary damages or enter into any settlement agreement containing material obligations;
- (v) incur any indebtedness for borrowed money or assume, guaranty, endorse or otherwise as an accommodation become responsible for the obligations of any other person, except for short-term borrowings made at prevailing market rates and terms consistent with prior practice;
 - (w) enter into any new material line of business;
- (x) engage in any material transaction or incur or sustain any material obligation not in the ordinary course of business consistent with past practice; and
 - (y) agree or make any commitment to take any actions prohibited by this Section 6.1.
- 6.2. <u>Affirmative Conduct Prior to Effective Time of the Merger</u>. Between the date hereof and the Effective Time of the Merger, Company shall cause Company and Company Subsidiaries to:
- (a) use their commercially reasonable efforts consistent with this Agreement to maintain and preserve intact their present business organization and to maintain and preserve their relationships and goodwill with customers, employees and others having business relationships with Company and Company Subsidiaries;
 - (b) use their commercially reasonable efforts to keep in full force and effect all of the existing material permits and licenses of Company;
- (c) use their commercially reasonable efforts to maintain insurance coverage at least equal to that now in effect on all properties which they own or lease and on their business operations;
 - (d) perform their contractual obligations and not become in default on any such obligations;

- (e) duly observe and conform to all applicable Laws;
- (f) maintain their assets and properties in good condition and repair, normal wear and tear excepted;
- (g) file all Tax Returns required to be filed with any tax authority in accordance with all applicable laws, timely pay all Taxes due and payable as shown in the respective Tax Returns that are so filed and ensure that the Tax Returns will, as of the time of filing, be based on tax positions that have substantial support under all applicable Laws; and
- (h) promptly notify Parent regarding receipt from any tax authority of any notification of the commencement of an audit, any request to extend the statute of limitations, any statutory notice of deficiency, any revenue agent's report, any notice of proposed assessment, or any other similar notification of potential adjustments to the Tax liabilities or attributes of Company, or any actual or threatened collection enforcement activity by any Tax authority with respect to tax liabilities of Company.
- 6.3. Access to Information. Company will afford, upon reasonable notice, to Parent and its representatives, counsel, accountants, agents and employees reasonable access during normal business hours to all of their business, operations, properties, books, files and records and will do everything reasonably necessary to enable Parent and its representatives, counsel, accountants, agents and employees to make a complete examination of the financial statements, business, assets and properties of Company and the condition thereof and to update such examination at such intervals as Parent shall deem appropriate. Such examination shall be conducted in cooperation with the officers of Company and in such a manner as to minimize any disruption of, or interference with, the normal business operations of Company. Upon the request of Parent, and upon Parent's execution and delivery of a customary waiver, Company will request E&Y to provide reasonable access to representatives of Parent, to auditors' work papers with respect to the business and properties of Company, including tax accrual work papers prepared for Company during the preceding 60 months, other than (a) books, records and documents covered by the attorney-client privilege, or that are attorneys' work product, and (b) books, records and documents that Company are legally obligated to keep confidential. All documents and information concerning Company so obtained from any of them (except to the extent that such documents or information are a matter of public record or require disclosure in the Proxy Statement or any of the public portions of any applications required to be filed with any Governmental Entity to obtain the approvals and consents required to effect the transactions contemplated hereby), shall be subject to the Confidentiality Agreement.
- 6.4. <u>Filings</u>. Company agrees that through the Effective Time of the Merger, Company's reports, proxy statements, registrations, statements and other filings required to be filed with any applicable Governmental Entity will comply in all material respects with all applicable Laws enforced or promulgated by the Governmental Entity with which it will be filed and none will contain any untrue statement of material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

- 6.5. Notices; Reports. Company will promptly notify Parent of any event of which Company obtains knowledge which has had or may have a Material Adverse Effect, or in the event that Company determines that it is unable to fulfill any of the conditions to the performance of Parent's obligations hereunder, as set forth in Articles 9 or 11 herein, and Company will furnish Parent (i) as soon as available, and in any event within one Business Day after it is mailed or delivered to the Board of Directors of Company or committees thereof, any report by Company for submission to the Board of Directors of Company or committees thereof, provided, however, that Company need not furnish to Parent communications of Company's legal counsel regarding Company's rights and obligations under this Agreement or the transactions contemplated hereby, or other communication or incident to Company's actions pursuant to Section 6.1(m) hereof, or books, records and documents covered by confidentiality agreements or the attorney-client privilege, or which are attorneys' work product, (ii) as soon as available, all proxy statements, information statements, financial statements, reports, letters and communications sent by Company to its shareholders or other security holders, and all reports filed by Company with the SEC or other Governmental Entities, and (iii) such other existing reports as Parent may reasonably request relating to Company.
- 6.6. Company Shareholders' Meeting. As promptly as practicable after the execution of this Agreement, Company will take action necessary in accordance with applicable law and its Articles of Incorporation and Bylaws to convene a meeting of its shareholders to consider and vote upon this Agreement and the transactions contemplated hereby so as to permit the consummation of the transactions contemplated hereby. The Board of Directors of Company shall recommend that its shareholders approve and adopt this Agreement and the transactions contemplated hereby, including the Merger; provided, however, that the Board of Directors of Company may withdraw, modify or change its recommendation to the shareholders if the Board determines in good faith, following consultation with outside legal counsel, that failure to do so would be inconsistent with its fiduciary duties under applicable law. Subject to the proviso of the immediately preceding sentence, Company will use its commercially reasonable efforts to obtain the requisite affirmative vote of the holders of the outstanding Company Stock for the approval and adoption of this Agreement and the Merger.
- 6.7. Proxy Statement. Company will promptly prepare or cause to be prepared the Proxy Statement, and further agrees to provide any information requested by Parent for the preparation of any applications necessary to consummate the transactions contemplated hereby. Company shall afford Parent a reasonable opportunity to review all such applications and all amendments and supplements thereto before the filing thereof. The Proxy Statement shall contain the favorable recommendation of the Board of Directors, provided however, that the Board of Directors of the Company may withdraw, modify or change the recommendation if the Board of Directors determines in good faith, following consultation with outside legal counsel, that to do so would be inconsistent with its fiduciary duties under applicable law. Company covenants and agrees that, with respect to the information relating to Company, the Proxy Statement will comply with all material applicable Laws, and will not contain any untrue statement of material fact or omit to state any material fact required to be stated therein or necessary to make the statements contained therein, in light of the circumstances under which they were made, not misleading. Company will use its commercially reasonable efforts to assist Parent in obtaining all approvals or consents of Governmental Entities necessary to effect the Merger and the transactions contemplated herein.

ARTICLE 7.

COVENANTS OF PARENT AND MERGER SUB

Parent and Merger Sub covenant and agree with Company as follows:

- 7.1. <u>Limitation on Conduct Prior to Effective Time of the Merger</u>. Between the date hereof and the Effective Time of the Merger, except as contemplated by this Agreement, each of Parent and its Subsidiaries shall not, without the prior written consent of Company, which consent Company shall not unreasonably withhold or delay:
- (a) take any action which would or is reasonably likely to (i) adversely affect the ability of Parent to obtain any necessary approvals of any Governmental Entity required for the transactions contemplated hereby; (ii) adversely affect Parent's ability to perform its covenants and agreements under this Agreement; or (iii) result in any of the conditions to the performance of Company's or Parent's obligations hereunder, as set forth in Articles 9, 10 or 11 herein not being satisfied; or
 - (b) agree or make any commitment to take any actions prohibited by this Section 7.1.
- 7.2. <u>Applications</u>. Parent will, as promptly as practicable, but in any event within 30 days of the date hereof, prepare and file in final form or cause to be prepared and filed in final form (it being recognized that applicable Governmental Entities may require supplemental filings after such filing in final form) any applications necessary to consummate the transactions contemplated hereby. Parent shall afford Company a reasonable opportunity to review all such applications (except for the confidential portions thereof) and all amendments and supplements thereto before the filing thereof.
- 7.3. <u>Notices; Reports</u>. Parent will promptly notify Company in the event that Parent determines that it is unable to fulfill any of the conditions to the performance of Company's obligations hereunder, as set forth in Articles 9 or 10 herein.
 - 7.4. <u>Indemnification and Directors' and Officers' Insurance</u>.
- (a) From and after the Effective Time of the Merger, the Surviving Corporation shall, to the fullest extent permitted by applicable Law, indemnify and hold harmless, and provide advancement of expenses to, each person who is now, or has been at any time prior to the date hereof or who becomes prior to the Effective Time of the Merger, an officer or director of Company (the "Indemnified Parties") against all losses, claims, damages, costs, expenses, liabilities or judgments or amounts that are paid in settlement of or in connection with any claim, action, suit, proceeding or investigation based in whole or in part on or arising in whole or in part out of the fact that such person is or was a director or officer of Company, and pertaining to any matter existing or occurring, or any acts or omissions occurring, at or prior to the Effective Time of the Merger, whether asserted or claimed prior to, or at or after, the Effective Time of the Merger (including matters, acts or omissions occurring in connection with the approval of this Agreement and the consummation of the transactions contemplated hereby)

("<u>Indemnified Liabilities</u>") to the same extent such persons are indemnified or have the right to advancement of expenses as of the date of this Agreement by Company pursuant to Company's Articles of Incorporation, bylaws and indemnification agreements, if any, in existence on the date hereof with any directors or officers of Company.

- (b) For a period of six years after the Effective Time of the Merger, the Surviving Corporation shall cause to be maintained in effect the current policies of directors' and officers' liability insurance maintained by Company (<u>provided</u> that the Surviving Corporation may substitute therefore policies with a substantially comparable insurer of at least the same coverage, amounts and retentions containing terms and conditions which are no less advantageous to the insured) with respect to claims arising from facts or events which occurred at or before the Effective Time of the Merger.
- (c) The Surviving Corporation shall pay all expenses, including reasonable fees and expenses of counsel, that an Indemnified Person may incur in enforcing the indemnity and other obligations provided for in this Section 7.4 to the extent that the Indemnified Person prevails in a dispute with the Surviving Corporation concerning the enforcement of this Section 7.4.
- (d) If the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other person and shall not be the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfers or conveys all or substantially all of its properties and assets to any person, then, and in each such case, to the extent necessary, proper provision shall be made so that the successors and assigns of the Surviving Corporation, as the case may be, shall assume the obligations set forth in this Section 7.4.
- (e) The provisions of this Section 7.4, (i) are intended to be for the benefit of, and shall be enforceable by, each Indemnified Party, his or her heirs and representatives and (ii) are in addition to, and not in substitution for, any other rights to indemnification or contribution that any such person may have by contract or otherwise.

ARTICLE 8.

ADDITIONAL COVENANTS

The parties hereto hereby mutually covenant and agree with each other as follows:

8.1. HSR Matters. Each party hereto shall make an appropriate filing of a Notification and Report Form pursuant to the HSR Act with respect to the transactions contemplated hereby as promptly as practicable after the date hereof. Each such filing shall request early termination of the waiting periods imposed by the HSR Act. Each party hereby agrees to use commercially reasonable efforts to cause a termination of the waiting period under the HSR Act without the entry by a court of competent jurisdiction of an order enjoining the consummation of the transactions contemplated hereby at as early a date as possible. Each party also agrees to respond promptly to all investigatory requests as may be made by the government. In the event that a Request for Additional Information is issued under the HSR Act, each party

agrees to furnish all information required and to comply substantially with such Request as soon as is practicable after its receipt thereof so that any additional applicable waiting period under the HSR Act may commence. Each party will keep the other party apprized of the status of any inquiries made of such party by the Department of Justice, Federal Trade Commission or any other governmental agency or authority or members of their respective staffs with respect to this Agreement or the transactions contemplated hereby. All filing fees to be paid by Parent, Merger Sub or Company in connection with filing Notification and Report Forms pursuant to the HSR Act shall be paid by Company. To the extent any other antitrust or similar notification or consent is required from any other Governmental Entity, such filings and costs shall be undertaken and borne by the Company.

8.2. <u>Insurance Approvals.</u>

- (a) As soon as reasonably practicable following the date of this Agreement, the Company and Parent shall cooperate with each other and use (and shall cause their respective Subsidiaries to use) their respective commercially reasonable efforts to prepare and file with relevant insurance regulators requests for approval of the transactions contemplated by this Agreement.
- (b) Parent shall give to the Company prompt written notice if it receives any material notice or other communication from any insurance regulator in connection with the transactions contemplated in this Agreement, and, in the case of any such written notice or communication, shall promptly furnish the Company with a copy thereof.
- (c) All of Parent's applications and substantive correspondence with the insurance regulators to the extent that such application or correspondence relates to an issue which, if the Merger were not consummated would be reasonably likely to have a material adverse effect on the Company which applications or correspondence shall be approved in advance by the Company (such approval not to be unreasonably withheld, conditioned or delayed).
- (d) The Company shall have the right to participate in and shall, to the extent practicable, receive reasonable prior notice of, all of Parent's telephone calls and meetings to the extent that an issue which, if the Merger were not consummated would be reasonably likely to have a Material Adverse Effect on the Company, is reasonably likely to be discussed and in which case the Company's right to participate shall be limited to discussions relating to such issue.
- 8.3. <u>Commercially Reasonable Efforts</u>. Subject to the terms and conditions of this Agreement, each party will use its commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable Laws to consummate the transactions contemplated by this Agreement as promptly as practicable.
- 8.4. <u>Public Announcements</u>. No press release or other public disclosure of matters related to this Agreement or any of the transactions contemplated hereby shall be made by Parent or Company unless the other party shall have provided its prior consent (which shall

not be unreasonably withheld, delayed or conditioned) to the form and substance thereof; provided, however, that nothing herein shall be deemed to prohibit any party hereto from making any disclosure which its counsel deems necessary or advisable in order to fulfill such party's disclosure obligations imposed by applicable Laws.

ARTICLE 9.

CONDITIONS PRECEDENT TO THE MERGER

The obligations of each of the parties hereto to consummate the transactions contemplated herein are subject to the satisfaction, on or before the Closing Date, of the following conditions:

- 9.1. <u>Shareholder Approval</u>. The Agreement and the transactions contemplated hereby shall have received all requisite approvals of the shareholders of Company.
- 9.2. <u>Insurance Approvals</u>. All insurance approvals of any Governmental Entity shall have been obtained and such approval shall be in full force and effect, and there shall be no proceeding, order or pending or threatened proceeding by insurance regulatory authorities in Minnesota, Michigan or Colorado, the results of such proceeding could materially restrict Surviving Corporation's ability to operate Company's business as it is currently being conducted in those states following the Closing date.
- 9.3. <u>No Judgments or Orders</u>. No judgment, decree, injunction, order or proceeding shall be outstanding by any Governmental Entity which prohibits or restricts the effectuation of, or threatens to invalidate or set aside, the Merger substantially in the form contemplated by this Agreement.
- 9.4. <u>HSR Approvals</u>. The waiting period (and any extension thereof) applicable to the consummation of the Merger under the HSR Act shall have expired or been terminated.
- 9.5. <u>Employment Agreements</u>. The following persons must have signed an Employment Agreement and are able to perform thereunder with Surviving Corporation in substantially the form reviewed and approved by Company's Board of Directors: Jeffrey B. Murphy; Keith D. Krueger; Patricia M. Sheveland; David M. Dietz; and Alfred L. LaTendresse.
- 9.6. <u>Claim & Claim Settlement Expense</u>. The Company's net ultimate liability for incurred loss and allocated loss adjustment expense (excluding inter-company loss adjusting expense) as determined in accordance with Statutory Accounting Policies for accident years 2006 and prior shall not have materially adversely changed from December 31, 2006, based on information available to Company on the Closing Date.
- 9.7. <u>Minnesota Workers' Compensation Assigned Risk Plan</u>. The Company's contract with the Minnesota Workers' Compensation Assigned Risk Plan shall be in full force and effect.

ARTICLE 10.

CONDITIONS PRECEDENT TO THE OBLIGATIONS OF COMPANY

All of the obligations of Company to effect the transactions contemplated hereby shall be subject to the satisfaction, on or before the Closing Date, of the following conditions, any of which may be waived in writing by Company:

- 10.1. Representations and Warranties; Performance of Covenants. All the covenants, terms and conditions of this Agreement to be complied with and performed by Parent or Merger Sub on or before the Closing Date shall have been complied with and performed in all material respects. Each of the representations and warranties of Parent contained in Article 5 of this Agreement shall have been true and correct in all respects on and as of the date of this Agreement and (except to the extent such representations and warranties speak as of an earlier date or for changes expressly contemplated by this Agreement) on and as of the Closing Date, subject to such exceptions as would not (individually or in the aggregate) have a material adverse effect on Parent and its Subsidiaries, taken as a whole, with the same effect as though such representations and warranties had been made on and as of the Closing Date (it being understood that, for purposes of determining the effect of such exceptions, all materiality qualifications contained in such representations and warranties shall be disregarded).
- 10.2. Officers' Certificate. There shall have been delivered to Company on the Closing Date a certificate executed by an officer of Parent and Merger Sub certifying, to the best of their knowledge, compliance with all of the provisions of Section 10.1.
- 10.3. <u>Employee Benefit Plans</u>. Company shall have received evidence reasonably satisfactory to it that all of Company's employee benefit plans, programs and arrangements have been treated as provided in Article 12 of this Agreement.

ARTICLE 11.

CONDITIONS PRECEDENT TO OBLIGATIONS OF PARENT AND MERGER SUB

All of the obligations of Parent and Merger Sub to effect the transactions contemplated hereby shall be subject to the satisfaction, on or before the Closing Date, of the following conditions, any of which may be waived in writing by Parent:

11.1. Representations and Warranties; Performance of Covenants. All the covenants, terms and conditions of this Agreement to be complied with and performed by Company at or before the Closing Date shall have been complied with and performed in all material respects. Each of the representations and warranties of Company contained in Article 4 of this Agreement shall have been true and correct in all respects on and as of the date of this Agreement and (except to the extent such representations and warranties speak as of an earlier date or for changes expressly contemplated by this Agreement) on and as of the Closing Date, subject to such exceptions as would not (individually or in the aggregate) have a Material Adverse Effect, with the same effect as though such representations and warranties had been made on and as of the Closing Date (it being understood that, for purposes of determining the

effect of such exceptions, all Material Adverse Effect and materiality qualifications contained in such representations and warranties shall be disregarded), except that the representations and warranties contained in Section 4.2(a) shall be true and correct in all respects except for insignificant differences at and as of the date hereof and the Closing date.

- 11.2. <u>Authorization of Merger</u>. All actions necessary to authorize the execution, delivery and performance of this Agreement by Company and the consummation of the transactions contemplated hereby shall have been duly and validly taken by the Board of Directors and shareholders of Company.
- 11.3. Officers' Certificate. There shall have been delivered to Parent on the Closing Date a certificate executed by the Chief Executive Officer and the Chief Financial Officer of Company certifying, to the best of their knowledge, compliance with all of the provisions of Sections 11.1 and 11.2.
- 11.4. <u>Company Dissenting Shares</u>. The number of shares of Company Stock which constitute Company Dissenting Shares shall not exceed 10% of the Company Stock issued and outstanding as of the Closing Date.
- 11.5. <u>Employee Benefit Plans</u>. Parent shall have received satisfactory evidence that all of Company's employee benefit plans, programs and arrangements have been treated as provided in Article 12 of this Agreement.
- 11.6. <u>Transaction Related Expenses</u>. Parent shall have received satisfactory evidence that the Transaction Related Expenses shall not have exceeded \$2,000,000.
- 11.7. No Material Adverse Effect. There shall not have occurred a Material Adverse Effect on the Company, ACIC or BCIC, nor any change, event or state of circumstances or facts shall have occurred that may be reasonably be expected to have a Material Adverse Effect on the Company, ACIC or BCIC.

ARTICLE 12.

EMPLOYEE BENEFITS

12.1. Employee Benefits.

(a) Following the Effective Time of the Merger, current employees of Company shall continue in the Company Employee Plans and Benefit Arrangements (other than equity-based plans or arrangements) or, in Parent's sole discretion, shall become eligible for the employee benefit plans and benefit arrangements of Parent (including, without limitation, the medical, dental, short and long term disability, life insurance, cafeteria plan, paid time off and 401(k) Plan, including any matching contributions) on substantially similar terms as such plans and arrangements are generally offered from time to time to employees of Parent in comparable positions with Parent, it being understood that, except as otherwise provided under the terms of any such Company Employee Plan or Benefit Arrangement, Parent shall be entitled from time to time to modify, terminate or supplement any such employee plans or benefit arrangements or to

substitute new employee plans or benefit arrangements for such employee plans or benefit arrangements in the exercise of its business judgment.

- (b) With respect to any employee plans and benefit arrangements of Parent in which any current employees of Company first become eligible to participate on or after the Effective Time of the Merger ("New Plans"), Parent shall (i) take commercially reasonable steps to cause the waiver of all pre-existing conditions, exclusions and waiting periods with respect to participation and coverage requirements under any such New Plans, (ii) recognize service of the employees of Company credited by Company prior to the Effective Time of the Merger for purposes of eligibility and vesting under the New Plans (and not for purposes of benefit accrual under any employee defined benefit pension or retiree medical plans), and to the extent permissible under such New Plans, and (iii) credit any deductibles, co-payments or other out-of-pocket expenses for the benefit plan year for each employee and dependent recognized or recognizable under the Company Employee Plans or Benefit Arrangements. Entitlement to Paid Time Off (PTO) of employees of Company accrued as of the Effective Time of the Merger shall not be reduced.
- (c) Parent agrees that immediately (but no later than 5 business days) after the Effective Time of the Merger, the Parent will cause the Surviving Corporation to pay or provide to those persons eligible to participate in the Company's 2007 Profit Sharing Plan (the "2007 Profit Sharing Plan") and employed by Company on the Effective Time of the Merger (and waiving any other employment related condition under the 2007 Profit Sharing Plan), (A) the cash incentive payment under Company's 2007 Profit Sharing Plan payable to such employee as if the last day prior to the Effective Time of the Merger were the last day of the performance period for purposes of calculating the amounts payable to eligible employees under the 2007 Profit Sharing Plan ("Adjusted Performance Period") and based upon the actual achievement of each measure of the 2007 Profit Sharing Plan for the Adjusted Performance Period compared to the respective fiscal year goals, such goals to be pro-rated based on the number of completed months in the Adjusted Performance Period.
- (d) Parent agrees that those persons who were employees of Company who remain as employees of Parent or the Surviving Corporation after the Effective Time of the Merger, on such date as determined by Parent in its sole discretion, will be entitled to participate in annual cash incentive performance programs generally offered to employees of Parent (to the extent that such programs are offered from time to time by Parent) on the same terms as such programs are generally offered from time to time to employees in comparable positions with Parent.
- (e) For a period until six months after the Effective Time of the Merger, Parent will cause the Surviving Corporation to pay and provide to employees of Company whose employment terminates on or after the Effective Time of the Merger severance benefits and other required benefits under conditions and in amounts that are no less favorable to the employee than those provided for under the Company's severance programs as in effect as of the Effective Time of the Merger. Any such severance and other benefits under the Company's severance programs shall be in addition to, and shall not be reduced or offset by any notice pay, severance pay or pay in lieu of notice required under any federal or state statute or regulation.

- (f) Parent will cause the Surviving Corporation to assume and perform all of the obligations of Company under the terms of any Executive Employment Agreement with any of the Company's executive officers, and to pay and provide to Employees of the Company whose employment terminates on or after the Effective Time if the Merger and who, as a result, become eligible for benefits under the terms of any Executive Employment Agreement, each and every benefit to which the employee is entitled under the terms of such agreements.
- (g) In the event Section 409A(a)(1)(B) of the Code requires a deferral of any payment to an employee who is a "specified employee" as that term is defined in Code 409A, such payment shall be accumulated and paid in a single lump sum on the earliest date permitted by Code 409A.
- 12.2. Company Stock Options and the Company Stock Option Plans. As soon as practicable following the date of this Agreement, the Board of Directors of Company (or, if appropriate, any committee administering the Company Stock Option Plans) shall take such actions as are required to: (i) provide that each outstanding Company Stock Option shall automatically accelerate so that each such Company Stock Option shall, immediately prior to the Effective Time of the Merger, become fully vested and fully exercisable for all the Shares at the time subject to such Company Stock Option and may be exercised by the holder thereof for any or all of such Shares as fully vested Shares, (ii) provide that the ESPP shall terminate prior to the earlier of the Effective Time of the Merger or December 31, 2007 and (iii) provide that, upon the Effective Time of the Merger, all outstanding Company Stock Options, to the extent not exercised immediately prior to the Effective Time of the Merger, shall be cancelled (and each holder of a Company Stock Option shall cease to have any rights with respect thereto except as provided in this Section 12.2(a)(ii)) in exchange for a cash payment by Company of an amount equal to (A) the excess, if any, of (x) the Merger Consideration over (y) the exercise price per share of Company Stock subject to such Company Stock Option, multiplied by (B) the number of shares of Company Stock Subject to such Company Stock Option.
 - (a) All amounts payable pursuant to this Section 12.2 shall be subject to any required withholding of taxes and shall be paid without interest.
- (b) The Board of Directors of Company (or, if appropriate, any committee administering the Company Stock Plans) shall adopt such resolutions or take such actions as are required to delete as of the Effective Time of the Merger the provision in any other Employee Plan or Benefit Arrangements of Company providing for the issuance, transfer or grant of any capital stock of Company or any interest in respect of any capital stock of Company and to ensure that following the Effective Time of the Merger no holder of a Company Stock Option or any participant in any Company Stock Plan or other Company Employee Plan or Benefit Arrangement shall have any right thereunder to acquire any capital stock of Company or the Surviving Corporation.

ARTICLE 13.

TERMINATION

- 13.1. <u>Termination</u>. This Agreement may be terminated at any time prior to the Effective Time of the Merger, whether before or after approval of this Agreement by the shareholders of Company, upon the occurrence of any of the following:
 - (a) By mutual agreement of Parent and Company, in writing;
- (b) By Parent or Company upon the failure of the shareholders of Company to give the requisite approval of this Agreement at the duly convened meeting of the shareholders of the Company required by Section 6.6, or any adjournment or postponement thereof; provided that the right to terminate this Agreement under this Section 13.1(b) shall not be available where the failure to obtain the requisite shareholder approval of this Agreement shall have been caused by the action or failure of the Company and such action or failure to act constitutes a breach by the Company of this Agreement:
- (c) By Company, upon written notice to Parent, if there shall have been a breach by Parent or Merger Sub of any of the covenants or agreements or any of the representations or warranties set forth in this Agreement on the part of Parent or Merger Sub, which breach, either individually or in the aggregate, would result in the failure of the condition set forth in Section 10.1 and which breach has not been cured within 30 days following written notice thereof to Parent or, by its nature, cannot be cured within such time period;
- (d) By Parent, upon written notice to Company, if there shall have been a breach by Company of any of the covenants or agreements or any of the representations or warranties set forth in this Agreement on the part of Company, which breach, either individually or in the aggregate, would result in the failure of the condition set forth in Section 11.1 and which breach has not been cured within 30 days following written notice thereof to Company or, by its nature, cannot be cured within such time period;
- (e) By Company or Parent if any conditions set forth in Article 9 shall not have been met by February 28, 2008; provided, however, that this Agreement shall not be terminated pursuant to this Section 13.1(e) if the relevant condition shall have failed due to the failure of the party seeking to terminate to comply in all material respects with its obligations under this Agreement;
- (f) By Company if any of the conditions set forth in Article 10 shall not have been met by February 28, 2008; provided, however, that this Agreement shall not be terminated pursuant to this Section 13.1(f) if the relevant condition shall have failed due to the failure of Company to comply in all material respects with its obligations under this Agreement;
- (g) By Parent if any of the conditions set forth in Article 11 shall not have been met by February 28, 2008; provided, however, that this Agreement shall not be terminated pursuant to this Section 13.1(g) if the relevant condition shall have failed due to the failure of Parent to comply in all material respects with its obligations under this Agreement;

- (h) By Parent if the Board of Directors of Company shall have failed to recommend adoption of this Agreement at the duly convened meeting of the shareholders of the Company required by Section 6.6, or withdrawn or modified or qualified in a manner adverse to Parent its favorable recommendation of this Agreement or recommended any Competing Transaction to the shareholders of Company; or
- (i) By Company if the Board of Directors of Company shall, concurrently with such termination, authorize Company to enter into an agreement with respect to a Competing Transaction; provided, however, that Company may only exercise its right to terminate this Agreement pursuant to this Section 13.1(i) if (i) Company shall have complied in all material respects with Section 6.1(m); (ii) the Board of Directors of Company, after consultation with its financial advisor, has reasonably determined in good faith that such Competing Transaction is a Superior Proposal (taking into account any proposal or offer which shall have been made by Parent to modify the terms of this Agreement); (iii) the Board of Directors of Company has reasonably determined in good faith (after consultation with outside legal counsel) that the failure to exercise such right of termination would be inconsistent with its fiduciary duties under applicable law; provided, that for purposes of this Section 13.1(i) the term "Competing Transaction" shall have the meaning set forth in Section 6.1(m), except that the reference to "15% or more" in the definition of Competing Transaction shall be deemed to be a reference to "a majority."

13.2. Effect of Termination.

- (a) In the event of termination of this Agreement by either Company or Parent as provided in Section 13.1, neither Company nor Parent shall have any further obligation or liability to the other party except under the terms of the Confidentiality Agreement, Section 13.1(i) and this Section 13.2; provided, however, that nothing herein shall relieve any party from liability for any willful and material breach of the warranties and representations made by it, or willful and material failure in performance of any of its covenants, agreements or obligations hereunder.
- (b) Company shall pay Parent an amount equal to the sum of \$1,000,000 (the "Termination Fee") if this Agreement is terminated as follows:
 - (i) If Parent or Company terminates the Agreement pursuant to 13.1(b) because of the failure of shareholders to approve the Agreement and within six months of such termination the Company has entered into a definitive agreement with a third party that had publicly announced its interest in pursuing a transaction with the Company prior to the Shareholders' Meeting at which this Agreement failed to receive approval, and such Termination Fee shall be due upon entry of a definitive agreement with a third party, provided, however, the Company shall not be required to pay the Termination Fee if the Company was otherwise entitled to terminate this Agreement pursuant to Sections 13.1(c), (e) or (f);
 - (ii) if Parent shall terminate this Agreement pursuant to Section 13.1(h), then Company shall pay Parent the Termination Fee on the Business Day following such termination; provided, however, the Company shall not be required to pay

the Termination Fee if the Company was otherwise entitled to terminate this Agreement pursuant to Section 13.1(c), (e) or (f); and

- (iii) if the Company terminates this Agreement pursuant to Section 13.1(i), the Company will pay the Termination Fee simultaneous with such termination.
- (c) If Company fails to pay all amounts due to Parent on the dates specified in this Section 13.2, then Company shall pay Parent interest on such unpaid amounts at the prime lending rate prevailing at such time, as published in the Wall Street Journal, from the date such amounts were required to be paid until the date actually received by Parent.

ARTICLE 14.

MISCELLANEOUS

- 14.1. Expenses. Except as otherwise provided herein, all Expenses incurred by Parent, Merger Sub and Company in connection with or related to the authorization, preparation and execution of this Agreement, the solicitation of shareholder approvals and all other matters related to the closing of the transactions contemplated hereby, including, without limitation of the generality of the foregoing, all fees and expenses of agents, representatives, counsel and accountants employed by either such party or its Affiliates, shall be borne solely and entirely by the party which has incurred the same. "Expenses" as used in this Agreement shall include all reasonable out-of-pocket expenses (including all fees and expenses of attorneys, accountants, investment bankers, experts and consultants to the party and its Affiliates) incurred by the party or on its behalf in connection with the consummation of the transactions contemplated by this Agreement.
- 14.2. <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party hereto to another shall be in writing and delivered personally or by confirmed facsimile transmission or sent by a recognized overnight courier service or by registered or certified mail, postage prepaid, with return receipt requested, addressed as follows:

To Parent or Merger Sub: Rockhill Holding Company

700 West 47th Street, Suite 350 Kansas City, Missouri 64112

Attention: President and Chief Executive Officer

Facsimile Number: (816) 412-7550

With a copy to: Lathrop & Gage 2345 Grand Avenue

Kansas City, Missouri 64108 Attention: Thomas H. Stahl

Facsimile Number: (816) 292-2001

To Company: RTW, Inc.

8500 Normandale Lake Blvd., Suite 1400

Bloomington, Minnesota 55437

Attention: President and Chief Executive Officer

Facsimile Number: (952) 893-3700

With a copy to: Lindquist & Vennum P.L.L.P.

80 South 8th Street Minneapolis, MN 55402

Attention: Thomas G. Lovett IV

Jonathan B. Levy

Facsimile Number: (612) 371-3207

Any such notice, request, instruction or other document shall be deemed received (i) on the date delivered personally or delivered by confirmed facsimile transmission, (ii) on the next Business Day after it was sent by overnight courier, delivery charges prepaid; or (iii) on the fourth Business Day after it was sent by registered or certified mail, postage prepaid. Any of the persons shown above may change its address for purposes of this section by giving notice in accordance herewith.

- 14.3. <u>Assignment</u>. All terms and conditions of this Agreement shall be binding upon and shall inure, to the extent permitted by law, to the benefit of the parties hereto and their respective permitted transferees and successors and permitted assigns; provided, however, that this Agreement and all rights, privileges, duties and obligations of the parties hereto, without the prior written approval of the other parties hereto, may not be transferred, assigned or delegated by any party hereto (by operation of law or otherwise) and any such attempted transfer, assignment or delegation shall be null and void.
- 14.4. <u>Counterparts</u>. This Agreement and any exhibit hereto may be executed in one or more counterparts, all of which, taken together, shall constitute one original document and shall become effective when one or more counterparts have been signed by the appropriate parties and delivered to each party hereto.
- 14.5. <u>Effect of Representations and Warranties</u>. The representations and warranties contained in this Agreement shall terminate immediately after the Effective Time of the Merger.
- 14.6. <u>Third Parties</u>. Each party hereto intends that this Agreement shall not benefit or create any right or cause of action for any person other than parties hereto, including, without limitation, current or former employees, directors, agents and independent, contractors, except as provided in Section 7.4(e). As used in this Agreement the term "parties" shall refer only to Parent, Merger Sub and Company as the context may require.
- 14.7. <u>Integration</u>. This Agreement constitutes the entire agreement between the parties pertaining to the subject matter hereof and supersedes all prior agreements and understandings of the parties in connection therewith.
- 14.8. <u>Knowledge</u>. Whenever any statement herein or in any list, certificate or other document delivered to any party pursuant to this Agreement is made "to the knowledge" or

"to the knowledge" of any party or another Person, such party or other Person shall make such statement based upon the actual knowledge after reasonable inquiry of the senior executive officers of such Person.

- 14.9. <u>Governing Law</u>. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Minnesota, regardless of the laws that might otherwise govern under applicable principles of conflict of laws thereof.
- 14.10. <u>Captions</u>. The captions contained in this Agreement are for convenience of reference only and do not form a part of this Agreement and shall not affect the interpretation hereof.
- 14.11. Severability. If any portion of this Agreement shall be deemed by a court of competent jurisdiction to be unenforceable, the remaining portions shall be valid and enforceable only if, after excluding the portion deemed to be unenforceable, the remaining terms hereof shall provide for the consummation of the transactions contemplated herein in substantially the same manner and with substantially the same effect as originally set forth at the date this Agreement was executed.
- 14.12. Waiver and Modification; Amendment. No waiver of any term, provision or condition of this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such term, provision or condition of this Agreement. Except as otherwise required by law, this Agreement, when executed and delivered, may be modified or amended by action of the Boards of Directors of Parent, Merger Sub and Company without action by their respective shareholders. This Agreement may be modified or amended or any provision hereof waived only by an instrument of equal formality signed by the parties or their duly authorized agents.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties to this Agreement have duly executed this Agreement as of the day and year first above written.

ROCKHILL HOLDING COMPANY

By: Terry L. Younghanz

Its: Chief Executive Officer

ROCKHILL ACQUISITION CORPORATION

By: Terry L. Younghanz
Its: Chief Executive Officer

RTW, INC.

By: Jeffrey B. Murphy
Its: President and Chief Executive Officer

<DOCUMENT>
<TYPE> EX-10.4
<FILENAME> c18876exv10w4.htm
<DESCRIPTION> Amendment to and Termination of 1995 Employee Stock Purchase Plan
<TEXT>

Termination of the Amended 1995 Employee Stock Purchase Plan On September 20, 2007, the Board of Directors of RTW, Inc. adopted the following resolutions

WHEREAS, the Company has established, and the shareholders have approved, the RTW, Inc. 1995 Employee Stock Purchase Plan ("ESPP"), which is intended to qualify under Section 423 and 421 of the Code, which ESPP was last amended by the Board on April 27, 2005 and the amendment was approved by the shareholders on June 15, 2005; and

WHEREAS, the 2007 Phase of the Plan would otherwise end as of April 15, 2008, at which time the Company would immediately exercise all options on behalf of any optionee who has not withdrawn from the ESPP and will issue to such optionee the number of whole shares of common stock purchased with the amount set aside on the optionee's behalf, and

WHEREAS, Section 12.2 of the Merger Agreement provides that the Company will terminate the 2007 Phase of the ESPP immediately prior to the earlier of the Effective Time of the Merger or December 31, 2007; and

WHEREAS, Section 3(b) of the ESPP reserves to the Committee the discretion to determine the length of time of each Phase of the ESPP;

THEREFORE, BE IT RESOLVED, that: (1) the Termination Date of the 2007 Phase of the ESPP be and hereby is the final payroll payment date immediately prior to the earlier of the Effective Time of the Merger or December 31, 2007; and (2) then outstanding options under the 2007 Phase of the ESPP will be immediately exercised as of the Termination Date for that whole number of Shares that may be purchased based on amount of each Participant's accumulated payroll withholdings as of the Termination Date, and that any amount not used to purchase Shares shall be returned to the Participant without interest; and (3) the Shares issued as a result of the exercise of the options under the ESPP be considered as outstanding as of the Effective Time of the Merger for purposes of receiving the Merger Consideration; and

FURTHER RESOLVED, that pursuant to the authority reserved to the Board under Section 12 of the ESPP, the ESPP be and hereby is amended to provide that no further Phases will commence after the Termination Date for the 2007 Phase; and

FURTHER RESOLVED, that, subject to the rights of the participants as set forth in the resolutions above, the ESPP be and hereby is terminated effective as of the Effective Date of the Merger, and that any and all common stock reserved under the ESPP as of the Effective Date of the Merger, shall be cancelled, withdrawn and no longer available for issuance under the ESPP.

<DOCUMENT>
<TYPE> EX-10.5
<FILENAME> c18876exv10w5.htm
<DESCRIPTION> Amendment to and Termination of 1994 Stock Plan
<TEXT>

Amendment to and Termination of the RTW, Inc. 1994 Stock Plan On September 20, 2007, the Board of Directors of RTW, Inc. adopted the following resolutions

WHEREAS, the Company has adopted, and the shareholders have approved the RTW, Inc. 1994 Stock Plan ("1994 Plan") to permit the Company to grant incentive and non-qualified options to purchase stock, stock appreciation rights, restricted stock and deferred stock to executives, key employees, and non-employee directors of the Company; and

WHEREAS, the Stock Option Agreements ("Stock Option") issued under the 1994 Plan incorporate by reference all of the terms of the Plan; and

WHEREAS, Section 12.2 of the Merger Agreement provides that: (a) each outstanding option shall automatically accelerate and be fully exercisable as of the Effective Time of the Merger (as defined in the Merger Agreement) without regard to any installment exercise thereof; and (b) all outstanding options as of the Effective Time of the Merger, to the extent not exercised immediately prior to the Effective Time, be cancelled in exchange for a cash payment equal to the spread between the exercise price of the option and the Merger Consideration times the number of unexercised options; and

WHEREAS, Section 5(c) of the 1994 Plan provides, in part, that: (a) notwithstanding anything in the 1994 Plan to the contrary, the Board has the authority and discretion to vary the exercise term of any Stock Option, if such action is deemed to be in the best interests of the Company; and (b) in the event of a merger of the Company, the Committee may, in its discretion, provide for one or more of the following actions: (i) the acceleration of the exercisability of any or all outstanding Stock Options; (ii) the complete termination of the 1994 Plan and cancellation of outstanding Stock Options not exercised prior to a date specified by the Board after a reasonable period of time in which to exercise vested Stock Options prior to the effectiveness of such merger;

THEREFORE, BE IT RESOLVED, that, pursuant to the authority under Section 5(c) of the 1994 Plan, each Stock Option issued under the 1994 Plan that is outstanding on the Effective Time be and hereby is amended to add the following:

"Notwithstanding anything in the Option Agreement to the contrary, in the event of the merger of the Company and another entity in which substantially all of the shareholders of the Company will receive consideration in a form other than shares of the successor or surviving entity: (a) the Option shall automatically accelerate and be fully exercisable as of the effective time of such merger without regard to any installment exercise thereof; and (b) if this Option is then outstanding on and remains unexercised as to any Shares as of the effective time of the merger, then this Option shall be cancelled as of such effective time and the Optionee shall receive, as the sole and complete consideration for the cancellation of the Option and in lieu of any and all interest in, and right to receive Shares of common stock of the Company or any substitute option or right in the common stock of its successor or survivor following the merger, with respect to all of the Shares subject to such cancelled Option as of the date of such cancellation, cash in an amount equal to the net amount, after applicable tax withholding, of: (i) the product of the excess, if any, of the merger consideration over the exercise price per Share (ii) multiplied by the number of Shares subject to the Option. If the exercise price per Share of this Option equals or exceeds the merger consideration, the cash amount shall be zero, and the Option shall be cancelled without further payment."

FURTHER RESOLVED, that immediately upon the Effective Time and the consummation of the transactions contemplated under the Merger Agreement, the 1994 Plan be and hereby is terminated, all outstanding options are cancelled, and no further shares shall remain available for issuance under the 1994 Plan, subject only to the payment as provided for above.

<DOCUMENT>
<TYPE> EX-10.6
<FILENAME> c18876exv10w6.htm
<DESCRIPTION> Amendment to and Termination of 2005 Stock Plan
<TEXT>

Amendment to and Termination of the RTW, Inc. 1994 Stock Plan On September 20, 2007, the Board of Directors of RTW, Inc. adopted the following resolutions

WHEREAS, the Company has adopted, and the shareholders have approved the RTW, Inc. 1994 Stock Plan ("1994 Plan") to permit the Company to grant incentive and non-qualified options to purchase stock, stock appreciation rights, restricted stock and deferred stock to executives, key employees, and non-employee directors of the Company; and

WHEREAS, the Stock Option Agreements ("Stock Option") issued under the 1994 Plan incorporate by reference all of the terms of the Plan; and

WHEREAS, Section 12.2 of the Merger Agreement provides that: (a) each outstanding option shall automatically accelerate and be fully exercisable as of the Effective Time of the Merger (as defined in the Merger Agreement) without regard to any installment exercise thereof; and (b) all outstanding options as of the Effective Time of the Merger, to the extent not exercised immediately prior to the Effective Time, be cancelled in exchange for a cash payment equal to the spread between the exercise price of the option and the Merger Consideration times the number of unexercised options; and

WHEREAS, Section 5(c) of the 1994 Plan provides, in part, that: (a) notwithstanding anything in the 1994 Plan to the contrary, the Board has the authority and discretion to vary the exercise term of any Stock Option, if such action is deemed to be in the best interests of the Company; and (b) in the event of a merger of the Company, the Committee may, in its discretion, provide for one or more of the following actions: (i) the acceleration of the exercisability of any or all outstanding Stock Options; (ii) the complete termination of the 1994 Plan and cancellation of outstanding Stock Options not exercised prior to a date specified by the Board after a reasonable period of time in which to exercise vested Stock Options prior to the effectiveness of such merger;

THEREFORE, BE IT RESOLVED, that, pursuant to the authority under Section 5(c) of the 1994 Plan, each Stock Option issued under the 1994 Plan that is outstanding on the Effective Time be and hereby is amended to add the following:

"Notwithstanding anything in the Option Agreement to the contrary, in the event of the merger of the Company and another entity in which substantially all of the shareholders of the Company will receive consideration in a form other than shares of the successor or surviving entity: (a) the Option shall automatically accelerate and be fully exercisable as of the effective time of such merger without regard to any installment exercise thereof; and (b) if this Option is then outstanding on and remains unexercised as to any Shares as of the effective time of the merger, then this Option shall be cancelled as of such effective time and the Optionee shall receive, as the sole and complete consideration for the cancellation of the Option and in lieu of any and all interest in, and right to receive Shares of common stock of the Company or any substitute option or right in the common stock of its successor or survivor following the merger, with respect to all of the Shares subject to such cancelled Option as of the date of such cancellation, cash in an amount equal to the net amount, after applicable tax withholding, of: (i) the product of the excess, if any, of the merger consideration over the exercise price per Share (ii) multiplied by the number of Shares subject to the Option. If the exercise price per Share of this Option equals or exceeds the merger consideration, the cash amount shall be zero, and the Option shall be cancelled without further payment."

FURTHER RESOLVED, that immediately upon the Effective Time and the consummation of the transactions contemplated under the Merger Agreement, the 1994 Plan be and hereby is terminated, all outstanding options are cancelled, and no further shares shall remain available for issuance under the 1994 Plan, subject only to the payment as provided for above.

<DOCUMENT>
<TYPE> EX-10.7
<FILENAME> c18876exv10w7.htm
<DESCRIPTION> Form of Employment Agreement
<TEXT>

EMPLOYMENT AGREEMENT

This EMPLOYMENT AGREEMENT (hereinafter the "Agreement") dated as of ____, 2007, is entered into by and between Rockhill Holding Company, a Delaware corporation, (the "Company"), and Jeffrey B. Murphy (the "Executive").

WHEREAS, the Company wishes to secure the services of the Executive; and

WHEREAS, the parties wish Executive to serve as the President and Chief Executive Officer of RTW, Inc., a Minnesota corporation ("RTW"), which is a wholly owned subsidiary of the Company, and to enter into this Agreement on the terms and conditions hereinafter set forth.

NOW, THEREFORE, in consideration of the premises and mutual covenants contained herein and for other good and valuable consideration, the receipt and adequacy of which are mutually acknowledged, the Company and the Executive agree as follows:

- 1. <u>Definitions</u>. For purposes of this Agreement, the following terms shall have the following meanings:
 - (a) "Base Salary" means the salary provided for in Section 4 or any increased salary granted to the Executive pursuant to Section 4.
 - (b) "Board" means the Board of Directors of the Company.
 - (c) "Cause" means the Executive:
 - (i) Has materially failed, neglected or refused to perform his duties under this Agreement;
 - (ii) Has engaged in malfeasance, misappropriation, fraud, dishonesty or gross misconduct in the performance of his duties to the Company and such act has the effect of injuring the business or reputation of the Company or any of its subsidiaries or affiliates;
 - (iii) Has violated any RTW or Company policies and practices and such violation has a material adverse effect on the Company or its subsidiaries or affiliates;
 - (iv) Has committed a material breach of this Agreement including those provisions relating to confidentiality, non-competition and non-solicitation set forth in Section 12 hereof; or
 - (v) Has been convicted of or plead guilty or no contest to a crime involving moral turpitude or a felony.
- (d) "Disability" means that the Executive (a) is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, or (b) is, by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, receiving income replacement benefits for a period of not less than 3 months under an accident and

health plan covering employees of the Executive's employer.

- (e) "Effective Date" means the date first indicated above.
- (f) "Good Reason" means that one or more of the following events shall have occurred:
- (i) The Company has materially changed or undermined the Executive's rights, duties and responsibilities as President and Chief Executive Officer of RTW or otherwise without the Executive's consent;
- (ii) The Company has relocated the executive offices of RTW outside of the Minneapolis, Minnesota metropolitan area or otherwise requires the Executive to conduct his business outside of the Minneapolis, Minnesota metropolitan area (except for the ordinary course of business travel as contemplated by Section 3(a) of this Agreement) without the consent of the Executive;
- (iii) The Company has committed a material breach of this Agreement, the Bonus Pool or any Award Agreement with Executive; or
- (iv) There has been a Change of Control. A Change of Control shall occur in the event of a change in the ownership of the Company which occurs on the date that any one person, or more than one person acting as a group, acquires ownership of stock of the Company that, together with stock held by such person or group, constitutes more than 50 percent of the total fair market value or the total voting power of the stock of the Company. Persons will be considered to be acting as a group if they are owners of a corporation that enters into a merger, consolidation, purchase or acquisition of stock, or a similar business transaction with the Company.
- (v) For the purpose of this Paragraph 1(f), the Executive must provide notice to the Company of the condition described in this Paragraph 1(f) giving rise to the Good Reason termination, and the Company will thereafter have a period of at least thirty days during which it may remedy the condition. In any event, the Executive must provide written notice to the Company of the Executive's resignation for Good Reason within not more than two years following the Change of Control.
- (g) "Party" or "Parties" means the Company and/or the Executive.
- (h) "Person" means any individual, corporation, partnership, limited liability company, joint venture, trust, estate, board, committee, agency, body, employee benefit plan or other person or entity.
- (i) "Proceeding" means any threatened or actual action, suit or proceeding, whether civil, criminal, administrative, investigative, appellate or other.
- (j) "Standard Benefit" means any amounts earned, accrued or owing to the Executive but not yet paid, and receipt of other benefits, if any, in accordance with, and subject to, the applicable plans and programs of the Company.
 - (k) "Term of Employment" means the period specified in Section 2.

2. Term of Employment.

- (a) The Company agrees to employ the Executive under this Agreement, and the Executive accepts such employment, for a period commencing on the Effective Date and ending on the second anniversary thereof, or such later date as provided in Section 2(b). Notwithstanding the foregoing, the Term of Employment shall be earlier terminated upon the termination of the Executive's employment, but only in accordance with the provisions of Section 9.
- (b) The Term of Employment shall be extended automatically for one additional year beginning on the last day before the second anniversary of the Effective Date (the "Extension Date") and on each anniversary of the Extension Date thereafter unless and until, not later than three months prior to the Extension Date or any anniversary of the Extension Date, the Company or the Executive gives written notice to the other party that the Term of Employment shall not be so extended.

3. Positions; Duties; Responsibilities; and Place of Employment.

- (a) During the Term of Employment, the Executive shall be employed as the President and Chief Executive Officer of RTW or in such other position or positions with the Company, its subsidiaries or affiliates, as the Board shall from time to time specify. The Executive, in carrying out his executive duties under this Agreement, shall report to the President and Chief Executive Officer of the Company. While employed by the Company hereunder, the Executive shall perform his duties at RTW's offices in Minneapolis, Minnesota; provided, however, that the Executive shall be required to travel as reasonably necessary in carrying out his duties and obligations hereunder. The Executive is required to work the hours and days reasonably necessary to fulfill his duties under this Agreement.
- (b) The Executive acknowledges that his duties and responsibilities hereunder will require his full business time and effort and agrees that, during the Term of Employment, he will not engage in any other business activity or have any business pursuits or interests that materially interfere or conflict with the proper performance of his duties hereunder.
- 4. <u>Base Salary</u>. Commencing as of the Effective Date, the Company shall pay the Executive an annualized Base Salary of \$350,000 during the Term of Employment. Such Base Salary shall be payable at intervals in accordance with the regular payroll practices of the Company applicable to executives, but no less frequently than monthly. The Board shall review the Base Salary no less frequently than annually during the Term of Employment; provided, however, that the Base Salary shall not be decreased during the Term of Employment below the amount set forth above without the Executive's consent (including, without limitation, for the purpose of determining benefits due under Section 9).
- 5. <u>Bonus Pool</u>. Commencing as of the Effective Date and terminating December 31, 2007, Executive shall continue to be eligible to participate in any existing RTW bonus plan. Beginning on January 1, 2008, the Executive shall be eligible to participate in the Rockhill Holding Company Bonus Plan during the remaining Term of Employment, subject to the terms and conditions of the Company's plan.

6. <u>Restricted Stock</u>. On January 1, 2008, the Executive shall be awarded 1,000 shares of restricted stock ("Restricted Shares") under the Rockhill Holding Company 2005 Restricted Stock Plan- Performance Vesting (the "Restricted Stock Plan"). Pursuant to the terms and subject to the conditions of the Restricted Stock Plan, the shares shall vest according to the achievement of certain performance criteria.

7. Other Benefits.

- (a) Employee Benefits. During the Term of Employment, the Executive shall be entitled to the benefits listed on Schedule A hereof and shall be eligible to participate in all employee benefit plans, programs and arrangements made available generally to the Company's executives in accordance with the terms and subject to the conditions of such plans, programs and arrangements; provided, however, that, except with respect to the benefits listed on Schedule A hereof, nothing in this Agreement shall be construed to require the Company to establish or maintain any such plans, programs or arrangements, or to prevent the Company from terminating any such plan, program or arrangement in accordance with its terms, except as required by law. From the period of close to a date specifically agreed to by the President and Chief Executive Officer of the Company and the Executive, said Executive shall continue to be eligible for benefits historically offered by RTW.
- (b) <u>Perquisites</u>. During the Term of Employment, the Executive shall participate in all fringe benefits and perquisites available to executives of the Company at levels and on terms and conditions that are commensurate with his positions and responsibilities at the Company. The Executive shall also receive such additional fringe benefits and perquisites as the Company may, in its discretion, from time to time elect to provide.
- (c) <u>Vacation, Holidays, and Leave</u>. During the Term of Employment, the Executive shall be entitled to vacation, holidays, and leave in accordance with the reasonable practices of the Company. In no instance shall said Executive lose any such or like days previously credited by RTW.
- 8. <u>Reimbursement of Business Expenses</u>. The Executive is authorized to incur reasonable expenses in carrying out his duties and responsibilities under this Agreement and the Company shall promptly reimburse the Executive for all such expenses, subject to documentation in accordance with reasonable policies of the Company.

9. Termination of Employment.

- (a) <u>Termination Due to Death</u>. If the Executive's employment hereunder is terminated due to his death, his estate or his beneficiaries (as the case may be) shall be entitled to the following:
 - (i) payment of Base Salary, in accordance with the Company's regular payroll practices (based on the Executive's rate of annual Base Salary at the time of his death), through the date of his death; and
 - (ii) payment of the Standard Benefit.
- (b) <u>Termination Due to Disability</u>. If the Executive's employment hereunder is terminated due to Disability, the Executive shall be entitled to the following:

- (i) payment of Base Salary, in accordance with the Company's regular payroll practices (based on the Executive's rate of annual Base Salary at the time of the Executive's termination of employment), through the date of the Executive's termination of employment until commencement of long-term disability payments, but in no event for more than one year following the last day of the Executive's employment;
 - (ii) payment of the Standard Benefit; and
- (iii) continued participation for one year for the Executive and each of his covered dependents in all medical, dental, hospitalization and life insurance coverages and in all other employee welfare benefit plans, programs and arrangements, in which the Executive and such dependents were participating at the time of the Executive's termination of employment, to the extent permitted by applicable law and the terms of such programs and arrangements, on terms and conditions no less favorable than those applying on such date.
- (c) <u>Termination by the Company for Cause</u>. The Company may terminate the Executive's employment for Cause at any time during the Term of Employment. If the Company terminates the Executive's employment for Cause, the Executive shall be entitled to the following:
 - (i) payment of Base Salary through the last day of the Executive's employment; and
 - (ii) payment of the Standard Benefit.
- (d) <u>Termination by the Company Without Cause or by the Executive for Good Reason</u>. The Company may terminate the Executive's employment without Cause and the Executive may terminate his employment voluntarily for Good Reason, at any time during the Term of Employment. If the Executive's employment hereunder is terminated by the Company without Cause or by the Executive for Good Reason, subject to Section 9(f), the Executive shall be entitled to:
 - (i) payment of Base Salary, in accordance with the Company's regular payroll practices (based on the Executive's rate of annual Base Salary at the time of the Executive's termination of employment) for twelve months following the last day of the Executive's employment;
 - (ii) continued participation for one year for the Executive and each of his covered dependents in all medical, dental, hospitalization and life insurance coverages and in all other employee welfare benefit plans, programs and arrangements, in which the Executive and such dependents were participating at the time of the Executive's termination of employment, to the extent permitted by applicable law and the terms of such programs and arrangements, on terms and conditions no less favorable than those applying on such date: and
 - (iii) payment of the Standard Benefit.

For the purposes of this Agreement and grants made to the Executive under the Restricted Stock Plan and prior to the second anniversary of the Effective Date, in the event the Term of Employment is not extended (whether at the Extension Date or any one

year anniversary thereof) as a result of the Company providing notice as provided in Section 2(b), such failure to extend shall be deemed a termination by the Company without Cause. For purposes of grants made to the Executive under the Restricted Stock Plan, termination by the Executive for Good Reason shall be deemed a termination by the Company without Cause.

- (e) <u>Voluntary Termination</u>. The Executive may voluntarily terminate his employment during the Term of Employment, provided he gives at least 30 days advance written notice. If the Executive voluntarily terminates his employment (and not because of Good Reason, his death or due to Disability), the Executive shall have the same entitlements hereunder as provided in Section 9(c) in the case of a termination by the Company for Cause.
- (f) Release. Notwithstanding any provision herein to the contrary, the Company may require that, prior to payment of any amount or provision of any benefit pursuant to Section 9(d)(i) or (ii), the Executive shall have executed a complete release of the Company and related parties in such form as is reasonably required by the Company and any waiting periods contained in such release shall have expired.

10. Assignability; Binding Nature.

- (a) This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors, heirs and personal representatives (in the case of the Executive) and assigns.
- (b) the Company's rights or obligations under this Agreement may be assigned or transferred by the Company only pursuant to a merger, consolidation or similar transaction in which the Company is not the continuing entity, or a sale or liquidation of all or substantially all of the assets and business of the Company; provided, that the Executive's written consent shall be required prior to the assignment or transfer of the Company's rights or obligations hereunder, and provided further, that the assignee or transferee is the successor to all or substantially all of the assets and business of the Company and such assignee or transferee assumes the liabilities, obligations and duties of the Company, as contained in this Agreement, either contractually or as a matter of law. In the event of any sale of assets and business or liquidation as described in the preceding sentence, the Company shall use its best efforts to cause such assignee or transferee to expressly assume the liabilities, obligations and duties of the Company hereunder and shall cause such assignee or transferee to deliver a legal, valid and enforceable written instrument in form and substance satisfactory to the Executive and his counsel to such
- (c) No rights or obligations of the Executive under this Agreement may be assigned or transferred by the Executive other than his rights to compensation and benefits, which may be transferred only by will or operation of law, or as provided in Section 15(e).
- 11. <u>Representations</u>. the Company represents and warrants that (a) it is fully authorized by action of the Board (and of any other Person whose action is required) to enter into this Agreement and to perform its obligations hereunder, and (b) upon the execution and delivery

of this Agreement by the Parties, this Agreement shall be the valid and binding obligation of the Company, enforceable against the Company in accordance with its terms.

12. Covenant Not to Compete; Confidentiality.

(a) Covenant Not to Compete.

- (i) The Executive agrees that for so long as the Executive is employed by the Company, and for a period of two years following the termination of the Executive's employment for any reason, the Executive shall not directly or indirectly:
 - (A) enter into or attempt to enter into a Restricted Business (as defined below) in the areas of the United States in which the Company or any of its affiliates or subsidiaries conduct business or in which the Company has planned to conduct business within two years thereafter, as a principal, partner, employee, consultant, agent, broker, intermediary, shareholder, investor, officer or director (other than as a holder of not in excess of 1% of the outstanding voting shares of any publicly traded company) which shall include, without limitation, the following activities: (1) providing risk management and or other services for any insurer or affiliate of an insurer in a Restricted Business; (2) developing, handling, or managing data or information concerning a Restricted Business; (3) participating in any decision, or developing or implementing any strategy, to engage in a Restricted Business; or (4) performing any functions in a Restricted Business that are the same as, or substantially similar to, the duties performed for the Company or any of its affiliates or subsidiaries at any time during the 12 months preceding the termination of the Executive's employment;
 - (B) induce or attempt to persuade any then-current employee, agent, manager, consultant or director of the Company or any of its affiliates or subsidiaries to terminate such employment or other relationship in order to enter into any Restricted Business with the Executive;
 - (C) use contracts, proprietary information, trade secrets, confidential information, customer lists, mailing lists, goodwill, or other intangible property used or useful in connection with the business of the Company or its affiliates or subsidiaries; or
 - (D) solicit or otherwise attempt to establish for the Executive or any other Person any business relationship with any Person which is, or during the one year period preceding the Executive's date of termination of employment was, a customer, client or distributor of the Company or any of its affiliates or subsidiaries.
- (ii) For the purposes of this Section 12, a "Restricted Business" is defined as (A) the business of producing, underwriting, writing and servicing insurance policies directly or through relationships with affiliated and unaffiliated insurance agents and agencies with respect to worker's compensation insurance

coverage that RTW, Inc. or its subsidiaries produced, had underwritten, written, or serviced during the Executive's Term of Employment, and (B) any other business that would be in direct competition with the business of the Company, RTW, or any affiliates or subsidiaries, as conducted on the date of the Executive's termination of employment.

- (iii) The covenants of the Executive set forth in this Section 12 shall be null and void and without any force or effect upon the effective date of any liquidation or dissolution of the Company.
- (iv) It is the desire and intent of the Parties that the provisions of this Section 12 shall be enforced to the fullest extent permissible under the laws and public policies applied in each jurisdiction in which enforcement is sought. Accordingly, if any particular portion of this Section 12 shall be adjudicated to be invalid or unenforceable, this Section 12 shall be deemed amended to delete therefrom the portion thus adjudicated to be invalid or unenforceable, such deletion to apply only with respect to the operation of this Section 12 in the particular jurisdiction in which such adjudication is made. The Executive acknowledges that he has received good and valuable consideration for the restrictive covenants contained in this Section 12.
- (b) <u>Confidentiality</u>. The Executive acknowledges that he will develop and be exposed to information that is or will be proprietary to the Company or its affiliates, including, but not limited to, customer lists, marketing plans, pricing data, product development plans and other intangible information. Such information shall be deemed confidential to the extent such information is not generally known to the public or in the Company's industry. The Executive agrees to use such information only in connection with the performance of his duties hereunder and to maintain such information in confidence; provided, however, that the Executive may disclose such information when required to by law or by a court, government agency, legislative body or other Person with apparent jurisdiction to order him to divulge, disclose or make accessible such information.
- (c) <u>Company Property</u>. Promptly following any termination of Executive's employment with the Company, the Executive shall return to the Company or its affiliates (as applicable) all property of the Company or its affiliates and all copies thereof in the Executive's possession or under his control.
- (d) <u>Non-Disparagement</u>. During the Term of Employment and thereafter for two years following the date of any termination of the Executive's employment with the Company, (i) neither the Executive nor the Company or any of its affiliates shall engage in conduct that could be disruptive in any way to the business or operations of the other or that could wrongfully interfere therewith, and (ii) neither the Executive nor the Company or any of its affiliates shall make at any time in the future any derogatory comments concerning the other or the business or operations of the other; provided, however, that nothing in this Section 12 (d) shall be deemed to prevent either Party from enforcing the other terms of this Agreement.
- 13. Governing Law and Arbitration; Waiver of Jury Trial. This Agreement shall be governed by the laws of the State of Minnesota, without reference to principles of conflicts or

choice of law under which the law of any other jurisdiction would apply. Any dispute or controversy arising under or in connection with this Agreement shall first be submitted to non-binding mediation. In the event such mediation fails to resolve such dispute or controversy then such dispute or controversy shall be settled exclusively by binding arbitration in St. Paul, Minnesota, in accordance with the rules of the American Arbitration Association then in effect, by an arbitrator that is acceptable to the Company and the Executive. In the event that the Company and the Executive do not agree on an acceptable arbitrator within 30 days of the date any claim for arbitration hereunder is asserted, the Company and the Executive each shall appoint one arbitrator, who shall appoint a third arbitrator to settle the dispute or controversy. In addition to any other awards granted to the prevailing party in any such dispute or controversy the prevailing party shall also be entitled to indemnification from the other party for all fees and expenses arising in connection with any such dispute or controversy governed by this Section 13 (including attorney's fees). Judgment may be entered on the arbitrator's award in any court having jurisdiction.

14. <u>Notices</u>. Any notice required or desired to be delivered under this Agreement shall be in writing and shall be delivered personally, by courier service, by registered mail, return receipt requested, or by telecopy and shall be effective upon actual receipt by the Party to which such notice shall be directed, and shall be addressed as follows (or to such other address as the Party entitled to notice shall hereafter designate in accordance with the terms hereof):

If to the Company:

Rockhill Holding Company 700 W. 47th St., Ste. 350 Kansas City, MO 64112 Attention: Terry L. Younghanz

If to the Executive, at his address as filed with the Company's personnel records.

15. Miscellaneous.

- (a) Entire Agreement. This Agreement contains the entire understanding and agreement between the Parties concerning the subject matter hereof and, as of the Effective Date, supersedes all prior agreements, understandings, discussions, negotiations and undertakings, whether written or oral, between the Parties with respect thereto. There is no applicable disciplinary or grievance procedure or collective bargaining agreement.
- (b) <u>Severability</u>. In the event that any provision or portion of this Agreement shall be determined to be invalid or unenforceable for any reason, in whole or in part, the remaining provisions of this Agreement shall be unaffected thereby and shall remain in full force and effect to the fullest extent permitted by law so as to achieve the purposes of this Agreement.
- (c) <u>Amendment or Waiver</u>. No provision in this Agreement may be amended unless such amendment is set forth in writing and signed by the Parties. No waiver by either Party of any breach of any condition or provision contained in this Agreement shall be deemed a waiver of any similar or dissimilar condition or provision at the same or any

prior or subsequent time. To be effective, any waiver must be set forth in writing and signed by the waiving Party.

- (d) <u>Headings</u>. The headings of the sections contained in this Agreement are for convenience only and shall not be deemed to control or affect the meaning or construction of any provision of this Agreement.
- (e) <u>Beneficiaries/References</u>. The Executive shall be entitled, to the extent permitted under any applicable law, to select and change a beneficiary or beneficiaries to receive any compensation or benefit hereunder following the Executive's death by giving the Company written notice thereof. In the event of the Executive's death or a judicial determination of his incompetence, reference in this Agreement to the Executive shall be deemed, where appropriate, to refer to his beneficiary, estate or other legal representative.
- (f) <u>Survivorship</u>. Notwithstanding anything contained herein to the contrary, if the Executive's employment with the Company terminates during the Term of Employment, Sections 1, 9, 12, 13, 14 and 15 of this Agreement, and the Parties' respective rights and obligations under such provisions, shall survive until all of the Parties' obligations under such provisions are satisfied.
- (g) <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which, when taken together, shall constitute one and the same instrument.
- (h) Withholding. The Company shall make deductions from any payments provided for herein in respect of payroll tax, as applicable, up to the maximum permitted by law, as well as any other amounts required to be withheld from time to time under any applicable income or employment tax laws or similar statutes or other provisions of law then in effect, and, with respect to any non-cash compensation or benefits with respect to which a tax withholding obligation will arise, may require as a condition to receipt of such compensation or benefit that the Executive make arrangements with the Company for the satisfaction of such tax withholding obligation.

ROCKHILL HOLDING COMPANY

IN WITNESS WHEREOF, the undersigned have executed this Agreement as of the date first set forth above.

By: Name: Title:	
EXECU	JTIVE
By: Name:	Jeffrey B. Murphy

$\underline{Schedule\;A}$

The Executive shall be entitled to the following benefits:

- 1. Term life insurance in the amount of \$2,000,000 with the Executive's spouse as beneficiary; and
- 2. Directors and Officers coverage in the amount of \$5,000,000.

<DOCUMENT>
<TYPE> EX-99.1
<FILENAME> c18876exv99w1.htm
<DESCRIPTION> Press Release
<TEXT>





FOR IMMEDIATE RELEASE

Contact: Terry Younghanz, President and CEO

Rockhill Insurance Company

(816).412-2801

Jeffrey B. Murphy, President and CEO Alfred L. LaTendresse, EVP and CFO

RTW, Inc. (952) 893-0403

RTW, Inc. Agrees to Merger with Rockhill Holding Company

Kansas City, Missouri and Minneapolis, Minnesota — September 21, 2007 — Rockhill Holding Company ("Rockhill) and RTW, Inc. ("RTW") (Nasdaq: RTWI), announced today that they have entered into a definitive merger agreement under which Rockhill will acquire all outstanding shares of RTW stock for \$12.45 per share in cash, in a transaction valued at approximately \$67.6 million. After the merger, RTW will continue to operate as a separate wholly-owned subsidiary of Rockhill.

The proposed transaction has been unanimously approved by the Boards of Directors of both companies and is subject to customary closing conditions, regulatory reviews and approvals and approval by the shareholders of RTW. The transaction is not subject to financing conditions. The parties expect the transaction to close on or before December 31, 2007.

"We believe we are putting RTW and its customers in very good hands," commented John O. Goodwyne, Chair of the RTW Board of Directors. "We are pleased that Rockhill will pay RTW shareholders a price per share that we believe is good value and provide immediate liquidity in a very thinly traded stock. The price of \$12.45 represents a premium of 54% percent over the Nasdaq market price on September 20, 2007 and is at a price higher than RTW stock has traded for at least the past five years. It has been a difficult market for small public companies, and we believe this transaction provides significant immediate shareholder value while enabling RTW to continue to provide quality service to its customers."

"We look forward to joining Rockhill," said Jeffrey B. Murphy, President and CEO of RTW, "and we are very excited about what Rockhill brings to us, including additional insurance expertise and access to capital. With this transaction, I believe RTW will be well-positioned to continue to serve its customers in the future. We are very proud of what we have accomplished as an independent company, building

strong operating insurance companies in American Compensation Insurance Company ("ACIC") and Bloomington Compensation Insurance Company ("BCIC") and adding service growth through RTW's Absentia® division. Rockhill is a great fit culturally for RTW and we expect little change in operations of the company as a result of the transaction. This is great news for our customers and employees."

"We are excited to add RTW to our portfolio of companies," said Terry Younghanz, President and CEO of Rockhill. "We have been searching for an organization that mirrors our specialty underwriting focus and helps us to diversify our existing operations. RTW is that company. Not only does RTW come to us with similar operating philosophies and cultural beliefs, but it is led by a very strong executive management team who has positioned the company very well for the future. We will bring capacity and capability to RTW as it continues to grow its workers' compensation insurance and service platforms into the future."

Advisors

Keefe, Bruette and Woods, Inc. acted as financial advisor and Lindquist & Vennum PLLP acted as legal counsel for RTW in this transaction. Lathrop & Gage L.C. acted as legal counsel to Rockhill in the transaction.

About Rockhill Holding Company

Rockhill Holding Company is a Kansas City based insurance holding company writing specialty property and casualty business through its two insurance company subsidiaries, Rockhill Insurance Company and Plaza Insurance Company.

About RTW

RTW, Inc., based in Minneapolis, Minnesota, is a leading provider of products and services to manage insured and self-insured workers' compensation, disability and absence programs. RTW provides these services, primarily directed at workers' compensation to: (i) employers insured through its wholly-owned insurance subsidiaries, ACIC and BCIC; (ii) self-insured employers on a fee-for-service basis; (iii) state assigned risk plans on a percent of premium basis; (iv) other insurance companies; and (v) agents and employers on a consulting basis, charging hourly fees. RTW developed two proprietary systems to manage disability and absence: (i) ID15®, designed to quickly identify those injured employees who are likely to become inappropriately dependent on disability system benefits, including workers' compensation; and (ii) RTW Solution®, designed to lower employers' disability costs and improve productivity by returning injured employees to work as soon as safely possible. RTW supports these proprietary management systems with state-of-the-art technology and talented people dedicated to its vision of transforming people from absent or idle to present and productive. ACIC writes workers' compensation insurance for employers primarily in Minnesota, Colorado and Michigan, but is growing in

new markets including Florida, Texas, Kansas, Connecticut, North Carolina and Iowa. BCIC offers workers' compensation insurance to selected employers in Minnesota and Colorado. In addition, through its Absentia[®] division, RTW has expanded and provides non-insurance products and service offerings nationally. RTW's services are effective across many industries. RTW, Inc. is traded on the Nasdaq Global Market under the symbol RTWI. For more information on RTW, Inc., please visit www.rtwi.com.

Safe Harbor Statement

Some of the statements made in this News Release, as well as statements made by RTW in periodic press releases and oral statements made by it to analysts and shareholders in the course of presentations about RTW, constitute "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. These forward-looking statements involve known and unknown risks, uncertainties and other factors that may cause RTW's actual results, performance or achievements to be materially different from any future results, performance or achievements expressed or implied by the forward-looking statements. Statements about RTW's future prospects are forward-looking and, therefore, involve certain risks and uncertainties. These and other risk factors are discussed in RTW's Report on Form 10-K for the year ended December 31, 2006 and from time to time in its other filings with the Securities and Exchange Commission.

In addition to these factors, forward looking statements in this press release are subject to (i) the ability of RTW and Rockhill to obtain the required regulatory approvals needed for consummation of the merger, and (ii) the approval by the shareholders of RTW.

Important Merger Information

In connection with the proposed merger with Rockhill Holding Company, RTW will file a proxy statement with the SEC. Shareholders and investors are advised to read the proxy statement when it becomes available because it will contain important information about the merger and RTW. Shareholders and investors may obtain a free copy of the proxy statement (when available) and other documents filed by the RTW with the SEC at the SEC web site at www.sec.gov. Free copies of the proxy statement, once available, and the Company's other filings with the SEC, may also be obtained from the Company at www.rtwi.com. Free copies of RTW's filings may be obtained by directing a written request to Secretary. RTW, Inc., 8500 Normandale Lake Blvd., Minneapolis, Minnesota 55439 Attention: Alfred LaTendresse or by telephone at 952-893-0403.

Participants in the Solicitation

RTW and its directors, executive officers and other members of its management may be deemed to be soliciting proxies from RTW's shareholders in favor of the merger. Investors and shareholder may obtain more detailed information regarding the direct and indirect interests in the merger of persons who may,

under the rules of the SEC, be considered participants in the solicitation of RTW's shareholders in connection with the merger by reading the preliminary and definitive proxy statements regarding the merger, which will be filed with the SEC. Information about RTW's directors and executive officers may be found in RTW's definitive proxy statement filed with the SEC on April 27, 2007. These documents will be available free of charge once available at the SEC web site at www.sec.gov or by directing a request to RTW as described above.

<DOCUMENT>
<TYPE> EX-99.2
<FILENAME> c18876exv99w2.htm
<DESCRIPTION> Quastion and Answer Fact Sheet
<TEXT>

To: All RTW Employees

From: Jeffrey B. Murphy, President and CEO

Attached for your reference are select questions and answers following the announcement and press release regarding the signing of a definitive agreement with Rockhill Holding Company to purchase RTW, Inc. (RTW or Company). The information contained in this Q&A document is intended to answer some of the questions on your mind at this point. This is not intended to be an all inclusive list of questions or answers. You should also know that there are some matters that legally we cannot discuss until an appropriate SEC filing is made. As you'll see, there are several questions that we cannot yet answer at this time. We hope that over the next few days and weeks, you will continue to raise questions and send them to me. I will collect all employee questions and ensure information is shared as soon as possible.

Proposed Acquisition of RTW, Inc. by Rockhill Holding Company Questions and Answer Fact Sheet September 26, 2007

General Questions:

Q: Who is Rockhill Holding Company?

A: Rockhill Holding Company (Rockhill) is privately held, headquartered in Kansas City, Missouri and backed by investors that have access to significant capital resources. Rockhill has been in operation since 2005 and is an insurance holding company serving specialty niches. Rockhill has two insurance subsidiaries: Rockhill Insurance Company (RIC) and Plaza Insurance Company (PIC). RIC is an Excess and Surplus lines insurer licensed in 48 states that writes niche specialty commercial business focused on wind exposures in the southeast U.S., earthquake exposures, residential contractors in various western states and commercial umbrella. PIC is an admitted insurance company licensed in 37 states that focuses on surety, personal and commercial business. RIC and PIC have a group A- Rating from A.M. Best.

Q: What are the basic financial terms of the proposed merger?

A: Rockhill is offering \$12.45 in cash for each outstanding share of RTW common stock and will pay to RTW stock option holders that amount per option share less the option exercise price. This price values the transaction at approximately \$67.6 million (including the value of stock options). There is no financing contingency in the announced merger.

The transaction is subject to customary terms and conditions including approval by Federal and State Regulators, RTW shareholders and antitrust clearance under the Hart-Scott Rodino Act. Additional details of the merger agreement will be disclosed in RTW filings with the Securities and Exchange Commission (SEC).

Q: What is the proposed timing for the transaction?

A: We anticipate that the transaction will close by December 31, 2007, subject to customary terms and conditions including approval by Federal and State Regulators, RTW shareholders and antitrust clearance under the Hart-Scott Rodino Act.

Q: What process do I go through to obtain cash for my shares?

A: This process will be explained in subsequent communications from RTW to its shareholders.

Q: When will RTW announce its results for the quarter ended September 30, 2007?

A: We intend to announce results for our third fiscal quarter ended September 30, 2007 on October 25, 2007.

Organizational Effects, Benefits & Pay Questions:

Q: How will the sale impact RTW employees?

A: RTW will operate as a standalone subsidiary of Rockhill. RTW believes Rockhill does not have any plans to eliminate any employees or functions within RTW and we expect little to change.

Q: What support is available if either me or my family are having a tough time with this change and need some support?

A: All RTW employees and their family members have access to a confidential Employee Assistance Program (EAP) through Midwest EAP Solutions. You can reach an EAP counselor at 1-800-383-1908. In addition, the executive team and management group are available to meet with you to discuss your concerns, normal reactions to change and how to best support yourself and others through change.

Q: What will happen to our existing benefit plans at RTW?

A: Many of the answers to the benefit questions are as yet unknown and need to be discussed by RTW and Rockhill. At a minimum, the RTW benefit plans will remain in place though the end of the year.

Q: Do our service dates transfer for benefits, vesting and severance purposes?

A: We do not have an answer at this time. Again, these will be answered as we discuss what is best for both organizations after closing.

Q: Are we still enrolling in the benefits plans in November as planned?

A: Yes, the RTW benefit plans will remain in place through at least the sale close date so it is important that you re-enroll in the RTW benefits plans as scheduled if you choose to continue coverage through RTW.

Q: Should this decision change how I think about enrolling in benefits in November-(i.e. flexible spending accounts, etc.)?

A: We believe it is in your best interest to proceed as if RTW were to remain a standalone company. We are researching this question further and will be prepared to address any special considerations during the open enrollment meetings expected in November.

Q: What happens to my PTO balance?

A: Any earned but unused PTO balance as of the date of the merger is considered earned and will not be reduced as a result of the sale of RTW but could be paid out with up to five days remaining in the accrual.

Q: Will merit increase salary reviews continue as scheduled?

A: Performance and merit increase reviews will continue as regularly scheduled until the sale close date. Managers and employees should work together to ensure that all reviews are completed on time. I expect that RTW will continue to follow its traditional pay change date of April 1 for 2008 and expect that we will align with the Rockhill schedule in 2009.

Incentive Plan Questions:

Q: Will we be eligible for a 2007 incentive payout?

A: The 2007 incentive payout will be based on company performance through the close date. The payment will be made within 5 business days following the sale close date. It is important to note that you must be an employee on the closing date to be eligible for this incentive.

Q: What can we be doing between now and close to maximize our incentive payout?

A: Work together to continue to focus on delivering our 2007 business plans and executing our growth objectives.

Stock Related Questions

Q: How will stock options be handled? What if a portion of my options are unvested?

A: Unvested stock options will be accelerated and vest immediately prior to the sale close date. At the time of closing, option holders will receive a cash payment, in lieu of company stock, equal to the number of shares underlying the option multiplied by the excess, if any, of the sales price over the exercise price. All amounts payable will be subject to any required taxes. If you are a stock option holder, you will receive further information and specific instructions regarding the exercise of your stock options.

- Q: Are there restrictions on exercising options, buying or selling stock prior to the close? Employees who are not RTW Officers can
- A: exercise and sell in accordance with the Company's Insider Trading Policy which can be referenced in the 2007 Team Member Handbook.

Q: How will the current Employee Stock Purchase Plan participation be handled for the current open period?

A: The plan will continue until the earlier of December 31, 2007 or the last payroll date prior to the closing date. No new enrollments or increases in deductions will be allowed. You continue to have the ability to decrease your deductions, stop your deductions or withdraw from the Plan.

Q: What if I own RTW stock, what happens to it at the time of the sale?

A: Instructions for redeeming your shares will be included in later communications from RTW to all its shareholders. The process for exchanging your shares in order to receive the merger consideration will be the same ones that apply for shareholders of RTW generally.

Integration Process Questions:

Q: How will the business be transitioned to the buyer?

A: RTW will be transitioned to Rockhill through a merger with an Acquisition Company recently created by Rockhill. Subsequent to closing and after the merger, RTW will be the surviving entity and will be a wholly-owned subsidiary of Rockhill. The corporate headquarters of RTW will remain in Bloomington, MN and all members of the executive management team will retain their previous positions. We will continue to build upon the brand name and identity that we have built over the past 15 years.

Communication and Q&A Process:

Q: How will we be kept informed of decisions?

A: Both RTW and Rockhill are committed to keeping you informed in a timely manner as information is available and decisions are made. We expect that we'll use a combination of meetings and written communications throughout the next few months.

Q: Where should I refer questions that I have that have not been answered?

A: Carol Strobl Allen and Jeff Murphy will be collecting all employee questions. Please send your questions via an e-mail or written note and we'll work to get you a response as soon as possible.

Q: How quickly should I expect to get a response to a question I have asked?

A: This will vary since we won't know all of the answers initially until specific decisions are made. Whenever possible we will answer questions within a few days even if it's letting you know that we don't know the answer yet.

Priorities / Focus

Q: How will this decision affect business until the sale is final?

A: Rockhill is expecting us to execute on our stated plans today, tomorrow and into the future. Continue to remain focused on delivering the exceptional results as you have done in the past.

Q: How will we fill current vacancies or resource needs between now and when the sale is final?

A: We will review each vacancy or resource need carefully and add personnel as necessary.

Q: Will there be any disruption in business?

A: Our job is to ensure there is minimal disruption in the business.

Q: What should I be working on or stop working on?

A: Generally, prioritize your efforts to focus on executing the 2007 plan and planning and execution of 2008. Specifically, please review your project list with your team leader or manager to agree to priorities.

Q: Will we still have quarterly financial meetings and earnings announcements?

A: Yes, until the sale is completed, we will continue our quarterly financial and reporting processes.

Q: Will we still be doing 2008 planning?

A: Yes. We will continue and complete the 2008 planning process.

Q: Should we still be working on new product development or new marketing or sales initiatives?

A: Yes. Continue work on all new product development, marketing and sales initiatives in the same manner as in the past.

Q: How should I handle a contract renewal that is scheduled to happen or be negotiated shortly?

A: Please discuss each of these situations directly with Jeff Murphy and/or Al LaTendresse. In general, we will not enter into any new contracts outside the normal course of business between now and the closing date.

Q: Should I still be spending as planned?

A: Please execute the approved plans. If you have a project that is not clearly a new product, 2007 execution, or necessary for 2008 plans, please review that with your department leader.

External Q & A

O: How will we communicate this to customers and partners?

A: Much of that work has already been done via phone calls. We plan to have all communication coordinated by our Underwriting and Marketing teams. All questions can be directed to the AUSM's, Account Managers, Director of Underwriting or Director of Marketing as they occur. Keith Krueger and Dave Dietz are the primary contacts coordinating our external communications. Please direct any non-customer outside questions to Jeff Murphy or Al LaTendresse.

Q: Will RTW's underwriting appetite or philosophy change after the merger with Rockhill?? We don't anticipate any changes in our

A: current underwriting appetite. Rockhill was looking for a company that seeks niche opportunities and provides value to the Workers' Compensation marketplace. We believe our approach meets those expectations.

Q: Will your approach towards reinsurance change?

A: RTW has maintained a conservative reinsurance approach to control volatility in our results, and at an attractive cost given current reinsurance conditions. While we don't anticipate any immediate changes in our program, over the long term we'll be evaluating options within reinsurance to optimize our financial performance in light of our new structure and capital base.

Q: RTW has a strong service reputation and orientation. Will that change?

A: We expect that our industry leading case and claims management processes will continue to improve. We are committed to strengthening our brand and image post merger and are constantly seeking new ways to improve our performance.

Q: Do you anticipate any changes in your agency appointment structure?

A: Our philosophy in agency appointment has always been centered on creating franchise value for our agents. We expect that value to increase after the sale is completed based on the financial strength of Rockhill and the analysis from A.M. Best. We will be looking to our agents to help build that value.

Q: Who will be my contact at RTW?

A: Our Agency Underwriting Sales Managers (AUSM's) will continue to be your main point of contact and access for company information. Our senior management will continue to be accessible to all insureds, agents and clients as well. We value our relationships and look forward to answering any questions you may have.

Important Merger Information

In connection with the proposed acquisition by Rockhill Holding Company, RTW will file a proxy statement with the SEC. **Shareholders and investors are advised to read the proxy statement when it becomes available because it will contain important information about the merger and the Company.** Investors may obtain a free copy of the proxy statement (when available) and other documents filed by the Company with the SEC at the SEC's web site at www.sec.gov. Free copies of the proxy statement, once available, and the Company's other filings with the SEC, may also be obtained from the Company at www.rtwi.com by clicking on the "Investors" tab and then following the link at "Financial Information" to "SEC Filings." Free copies of the Company's filings may be obtained by directing a written request to RTW, Inc., 8500 Normandale Lake Boulevard, Suite 1400, Bloomington, MN 55437, Attention: Alfred L. LaTendresse or by telephone at 952-893-0403.

Participants in the Solicitation

The Company and its directors, executive officers and certain other members of its management may be deemed to be soliciting proxies from the Company's stockholders in favor of the merger. Investors and stockholders may obtain more detailed information regarding the direct and indirect interests in the merger of persons who may, under the rules of the SEC, be considered participants in the solicitation of the Company's stockholders in connection with the merger by reading the preliminary and definitive proxy statements regarding the merger, which will be filed with the SEC. Information about the Company's directors and executive officers may be found in the Company's definitive proxy statement filed with the SEC on April 27, 2007. These documents, once available, will be available free of charge at the SEC's web site at www.sec.gov or by directing a request to the Company as described above.